



# Review Recent Techniques Used in Home Energy Management Systems: A Review

Isaías Gomes <sup>1,2,3</sup>, Karol Bot <sup>1</sup>, Maria Graça Ruano <sup>1,4,\*</sup> and António Ruano <sup>1,2</sup>

- <sup>1</sup> Faculty of Science & Technology, University of Algarve, 8005-294 Faro, Portugal; ilgomes@ualg.pt (I.G.); kbot@ualg.pt (K.B.); aruano@ualg.pt (A.R.)
- <sup>2</sup> IDMEC, Instituto Superior Técnico, Universidade de Lisboa, 1950-044 Lisboa, Portugal
- <sup>3</sup> ICT, University of Évora, 7002-554 Evora, Portugal
- <sup>4</sup> CISUC, University of Coimbra, 3030-290 Coimbra, Portugal
- Correspondence: mruano@ualg.pt

**Abstract:** Power systems are going through a transition period. Consumers want more active participation in electric system management, namely assuming the role of producers–consumers, prosumers in short. The prosumers' energy production is heavily based on renewable energy sources, which, besides recognized environmental benefits, entails energy management challenges. For instance, energy consumption of appliances in a home can lead to misleading patterns. Another challenge is related to energy costs since inefficient systems or unbalanced energy control may represent economic loss to the prosumer. The so-called home energy management systems (HEMS) emerge as a solution. When well-designed HEMS allow prosumers to reach higher levels of energy management, this ensures optimal management of assets and appliances. This paper aims to present a comprehensive systematic review of the literature on optimization techniques recently used in the development of HEMS, also taking into account the key factors that can influence the development of HEMS at a technical and computational level. The systematic review covers the period 2018–2021. As a result of the review, the major developments in the field of HEMS in recent years are presented in an integrated manner. In addition, the techniques are divided into four broad categories: traditional techniques, model predictive control, heuristics and metaheuristics, and other techniques.

**Keywords:** home energy management system; heuristics; metaheuristics; model predictive control; MILP

# 1. Introduction

In the past, energy management systems have been specifically designed for energy production and distribution/commercialization companies with the aim of monitoring and optimizing energy flows in the power system. More recently, information and communication technologies have played an important role in the energy sector [1], enabling the development of this type of management system at the end-user level. In this context, the energy end-users experienced a changing role over recent years, from a passive to an active perspective [2]. The initial objective was to manage energy consumption in their homes by the end-user themselves. These energy management systems were then called home energy management system (HEMS). At this stage, the main interest of end-users who have a HEMS is to reduce the value of the electricity bill. In a next phase, in addition to reducing electricity costs, end consumers began to worry about energy efficiency and the possibility of producing their own energy, mainly from renewable sources. Thus, currently, HEMS can be considered as a true energy hub, enabling the management of both the energy produced and the energy consumed.

Consequently, HEMSs are of great interest to the public, but particularly to the scientific community, with the aim of providing more and more management facilities, involving innovative techniques with the possibility of achieving greater rigor in this management



Citation: Gomes, I.; Bot, K.; Ruano, M.G.; Ruano, A. Recent Techniques Used in Home Energy Management Systems: A Review. *Energies* **2022**, *15*, 2866. https://doi.org/10.3390/ en15082866

Academic Editors: Angelo Zarrella and Kian Jon Chua

Received: 8 February 2022 Accepted: 12 April 2022 Published: 14 April 2022

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). process. Additionally, the ability to deal with a large volume of data in this area is challenging for the scientific community [3].

In this sense, this paper aims to contribute to the development of HEMS, presenting a systematic review of the techniques used in the design of HEMS and serving as a complement to other review works previously presented in this field. Within the scope of techniques, a complete analysis of the techniques used in different categories is undertaken, such as traditional techniques, model predictive control, heuristics and metaheuristics, and other techniques. This paper also presents the main computational and technical issues involved in HEMS design. To the best of our knowledge, this is the first article to carry out a systematic review addressing in an integrated way the techniques, computational issues, and technical issues considered in the design of HEMS.

The rest of the manuscript is organized as follows: Section 2 describes the systematic review rationale; Section 3 presents the techniques used in home energy management systems; Section 4 points out the related computational issues; Section 5 deals with the technical issues considered in HEMS; Section 6 briefly extends the review to energy communities and microgrids; and Section 7 provides a discussion and draws some conclusions. The number of citations and the origins of the papers are shown in Appendix A.

## 2. Systematic Review Rationale

To contribute to the development of innovative strategies for HEMS, this study consists of a systematic review of the applications of the techniques used in HEMS and the associated technical and computational issues. The review follows the guidelines proposed by Kitchenham [4] for a systematic review. According to these guidelines, at the time of pre-review it is necessary to define the research questions and a review protocol, i.e., the plan on which all review procedures are based. Thus, in planning the review stage, in addition to the research questions, the following must be defined: the search strategy, including the search words (keywords); the eligibility of the data, with the definition of inclusion and exclusion criteria. Then, in the conducting the review stage the review starts and should include data collection and data analysis. Finally, the results are presented, that is, the detailed analysis of the papers included in the work.

#### 2.1. Planning the Review

Having a plan is one of the most important steps in producing a good quality review article, as it allows to avoid mistakes that can slow down the process and reduce the quality of the review. Defining research questions is the most important procedure at this stage. Additionally, the plan should include the search strategy and data eligibility.

## 2.1.1. Research Questions

To achieve the objectives of the review, it is essential to be able to concretely define the subject to be addressed. This is possible through so-called research questions.

Formally, the research questions for this review can be defined as:

- 1. What are the main developments in terms of optimization techniques within the scope of HEMS in recent years?
- 2. What are the key factors at a technical and computational level that must be considered in the development of HEMS?

To address research question 1, the main methods and techniques used in the development of HEMS were surveyed. The techniques were grouped into the following broad categories: mathematical programming, heuristics and metaheuristics, model predictive control (MPC), and other techniques.

To address research question 2, a comprehensive review of the technical and computational aspects that are considered in the development of HEMS was performed. Among the aspects considered are the problem size, the resolution, the planning horizon, and the time steps, as well as other important considerations such as user behavior, demand-side management, uncertainty, and the consideration of multi-objective problems.

#### 2.1.2. Search Strategy

The search process is based on the identification of articles on the subject to be treated and should have as its fundamental objective the need to answer the research questions, highlighted above. In this process, only publications in the database Web of Science from the year 2018 to 2021 were considered.

#### Keywords

Search terms were defined considering the most representative words that meet the objectives of this review and that are related to HEMS. Preliminary searches were performed and terms that did not add articles were discarded. The following terms were combined, obtaining the so-called "search string" [4]:

(("home" or "residential" or "building" or "household" or "domestic") AND ("energy management" or "power management" or "energy scheduling" or "power scheduling" or "optimal energy" or "optimal scheduling")) OR ("HEMS").

This is the search string that was introduced in the database to obtain the results to the queries.

# 2.1.3. Eligibility of the Data

To define the eligibility of articles to be included in this review, some exclusion and inclusion criteria were considered from the initial set of articles obtained by the research.

# Inclusion Criteria

- 1. Articles that provide a clear technique for HEMS.
- 2. Articles addressing computational performance issues when developing HEMS.
- 3. Articles that address essential technical issues of HEMS.
- 4. Articles written in English, available for free, or available on platforms available to the research team.

# **Exclusion** Criteria

- 1. Articles not related to techniques for HEMS.
- 2. Articles that exclusively address thermal comfort management in HEMS.
- 3. Articles published before 2018.

Articles that met at least one of the criteria were removed from the analysis.

# 2.2. Conducting the Review

At this stage, the actual review begins. Search results and important information about data collection and data analysis were presented.

# Data Collection and Analysis

The number of articles to be considered as data collection, how many were included, and how many were excluded from the collection must be quantified. To analyze the data and gather relevant information for a systematic review, some attributes were defined:

- Authors, title, year of publication, and journal/conference.
- Techniques.
- Computational issues.
- Technical issues.

The data were organized in tables and figures to allow an easier and faster analysis, always aiming at answering the research questions.

The number of selected papers according to the inclusion/exclusion criteria is shown in Table 1.

Reference

[5]

Database	Search -	Round 1		Round 2	
		Crit	teria	Crit	teria
Web of Science	569	Included 306	Excluded 263	Included 101	Excluded 205

Table 1. Number of selected articles according to the inclusion/exclusion criteria.

In round 1, the titles and abstracts of the papers were analyzed in detail, considering the inclusion and exclusion criteria, resulting in a total of 306 papers. Then, in round 2, the pre-selected papers were analyzed in detail to check the completion of the inclusion and exclusion criteria. Namely, in round 2 it was analyzed whether each paper explicitly focused on techniques applied in HEMS, being discarded if not. As a result, 101 papers were obtained.

The number of selected articles according to article type is shown in Figure 1.



Figure 1. Number of selected articles according to article type.

The previous review articles and their main features are shown in Table 2. Compared with the reviews mentioned in Table 2:

- (i) The present work is focused on residential EMS, addressing a broad range of technical and computational issues specifically assessed from a residential point of view, facilitating understanding of the employed methods and modeling setups during the reading.
- (ii) Only references presenting a clear and extensive description of the employed methods were included in this review.

faced by HEMS.

Title	Main Features	
Home energy management system (HEMS): concept, architecture, infrastructure, challenges and energy	Focus on concepts, architecture, infrastructure, and energy management scheme of HEMS. Discussion of goals and challenges	

Table 2. Review articles and main features since 2018.

management schemes

[6]	A survey on home energy management	Focus on concept and components of HEMS, electric vehicle integration, demand-side management, appliances categorization, discussion of uncertainty incorporation, scheduling techniques, and security.
[7]	Optimization methods for power scheduling problems in smart home	Focus on the concepts of smart grid, smart homes, and mainly scheduling of smart home appliances. The scheduling techniques are divided in exact algorithms and metaheuristic algorithms. Includes a discussion of various types of demand response programs and price schemes.

Reference	Title	Main Features
[8]	Smart home energy management system—a review	Focus on the concept of smart HEMS and its architecture. Explicit demonstration of formulations for HEMS. Discussion of optimization techniques and solution methods, forecasting, and energy trading and tariffs. Discussion and analysis of factors in HEMS.
[9]	A review of reinforcement learning for autonomous building energy management	Exclusively related to review of reinforcement learning for building energy management.
[10]	A Review of Deep Reinforcement Learning for Smart Building Energy Management	Exclusively related to the applications of deep reinforcement learning for EMS.
[11]	Home Energy Management System Concepts, Configurations, and Technologies for the Smart Grid	Focus on main concepts, configurations, enabling technologies, and communication technologies.
[12]	Review on Home Energy Management System Considering Demand Responses, Smart Technologies, and Intelligent Controller	Focus on previous and current studies focusing on HEMS from a demand response perspective.
[13]	A Review of Internet of Energy (IoE) Based Building Energy Management Systems: Issues and Recommendations	Reviews the use of IoE for improving building EMS performance and its associated technologies.
[14]	Of impacts, agents, and functions: An interdisciplinary meta-review of smart home energy management systems research	EMS benefits for the energy systems.
[15]	Intelligent Controllers and Optimization Algorithms for Building Energy Management Towards Achieving Sustainable Development: Challenges and Prospects	Focus on optimization objectives for comfort, energy use, and scheduling in EMS.
[16]	Application of DR and co-simulation approach for renewable integrated HEMS: a review	Reviews the feasibility of HEMS methods according to their control techniques and functionalities.
[17]	A review of strategies for building energy management system: Model predictive control, demand side management, optimization, and fault detect and diagnosis	Strategies for EMS focusing on model predictive control, optimization, energy efficiency, fault detection, and diagnosis

 Table 2. Cont.

In addition to the papers presented above, we strongly recommend the reader to consult the article [18] which is one of the most cited and most complete papers in the HEMS review.

Journals with at least two articles in the final selection are shown in Table 3.

Publisher	Number of Articles
IEEE	12
MDPI	9
Elsevier	6
Elsevier	5
IEEE	5
	Publisher IEEE MDPI Elsevier Elsevier IEEE

Journal	Publisher	Number of Articles
Sustainable Cities and Society	Elsevier	5
Electric Power Systems Research	Elsevier	4
Electronics	MDPI	3
IEEE Internet of Things Journal	IEEE	3
IEEE Transactions on Industrial Informatics	IEEE	3
International Journal of Electrical Power and Energy Systems	Elsevier	3
CSEE Journal of Power and Energy Systems	CSEE	2
International Journal of Renewable Energy Research	Gazi University	2
International Transactions on Electrical Energy Systems	Wiley	2
Journal of Building Engineering	Elsevier	2
Journal of Energy Storage	Elsevier	2

The number of selected papers distributed by year of publication is shown in Figure 2.



Figure 2. Number of selected articles distributed by publication years.

## 3. Techniques Used in Home Energy Management Systems

HEMS are complex systems, and the decision maker must consider two fundamental issues when planning the system operation: firstly, the system modeling, and secondly, the techniques used to obtain a solution to the problem. The problems on which HEMS are based can generally be formulated as linear programming problems, non-linear programming, and its variants, mixed-integer linear programming (MILP), and non-linear integer programming problems. In general, the problem of planning the operation of systems such as a home is a non-linear problem with a significant number of constraints and variables. However, through function linearization techniques it is possible to obtain approximations that turn the problem into a linear problem. A significant number of technique chosen depends on the nature of the problem at hand. Generally, these techniques can be divided into the following categories, adopted in this work for further classification: traditional techniques, model predictive control, heuristics and metaheuristics, and other techniques, as shown in Figure 3.

Table 3. Cont.



Figure 3. Adopted classification of the HEMS techniques.

The number of selected articles according to these categories is shown in Table 4.

Table 4. Number of selected articles according to the category of techniques.

Category	Number of Articles
Traditional techniques	31
Model predictive control	8
Heuristics and metaheuristics	32
Other techniques	15

The category, technique, and simulation platform of the articles are shown in Table 5.

Reference	Category	Technique	Simulation Platform
[19]	Traditional techniques	Linear programming	MATLAB
[20]	Traditional techniques	Mixed-integer linear programming	MATLAB
[21]	Traditional techniques	Mixed-integer linear programming	-
[22]	Traditional techniques	Mixed-integer linear programming	-
[23]	Traditional techniques	Mixed-integer linear programming	MATLAB
[24]	Traditional techniques	Mixed-integer linear programming	GAMS
[25]	Traditional techniques	Mixed-integer linear programming	IBM ILOG CPLEX optimization studio
[26]	Traditional techniques	Mixed-integer linear programming	MATLAB
[27]	Traditional techniques	Non-linear programming	MATLAB
[28]	Traditional techniques	Mixed-integer non-linear programming	AIMMS
[29]	Traditional techniques	Mixed-integer non-linear programming	MATLAB
[30]	Traditional techniques	Dynamic programming	MATLAB
[31]	Traditional techniques	Dynamic programming	-
[32]	Traditional techniques	Stochastic programming	GAMS
[33]	Traditional techniques	Stochastic programming	GAMS
[34]	Traditional techniques	Stochastic programming	-
[35]	Traditional techniques	Stochastic programming	MATLAB
[36]	Traditional techniques	Stochastic programming	GAMS
[37]	Traditional techniques	Stochastic programming	Java
[38]	Traditional techniques	Stochastic programming	GAMS
[39]	Traditional techniques	Stochastic programming	GAMS
[40]	Traditional techniques	Stochastic programming	

Table 5. Selected papers by category, technique, and simulation platform.

# Table 5. Cont.

Reference	Category	Technique	Simulation Platform
[41]	Traditional techniques	Stochastic programming	GAMS
[42]	Traditional techniques	Robust programming	Python
[43]	Traditional techniques	Robust programming	GAMS
[44]	Traditional techniques	Robust programming	GAMS
[45]	Мо	del predictive control	-
[46]	Мо	del predictive control	-
[47]	Мо	Model predictive control	
[48]	Мо	del predictive control	MATLAB
[49]	Мо	del predictive control	MATLAB
[50]	Heuristics and metaheuristics	Grey wolf optimization, binary particle swarm optimization, genetic algorithm, and wind-driven optimization	MATLAB
[51]	Heuristics and metaheuristics	Genetic algorithm	MATLAB
[52]	Heuristics and metaheuristics	Natural aggregation algorithm	-
[53]	Heuristics and metaheuristics	Bat algorithm, grey wolf optimization, moth flam optimization, and Harris hawks optimization	MATLAB
[54]	Heuristics and metaheuristics	Integrated multi-objective antlion optimization, multi-objective particle swarm optimization, the second version of the non-dominated sorting genetic algorithm, and the basic antlion optimizer algorithm	MATLAB
[55]	Heuristics and metaheuristics	Hybrid harmony search algorithm and particle swarm optimization	-
[56]	Heuristics and metaheuristics	Real coded genetic algorithm	-
[57]	Heuristics and metaheuristics	Hybrid grey wolf genetic algorithm	MATLAB
[58]	Heuristics and metaheuristics	Polar bear optimization	-
[59]	Heuristics and metaheuristics	Particle swarm optimization	-
[60]	Heuristics and metaheuristics	Grey wolf optimization	MATLAB
[61]	Heuristics and metaheuristics	Improved genetic algorithm	-
[62]	Heuristics and metaheuristics	Genetic algorithm	MATLAB
[63]	Heuristics and metaheuristics	Grey wolf optimization	-
[64]	Heuristics and metaheuristics	Enhanced leader particle swarm optimization	MATLAB
[65]	Heuristics and metaheuristics	Hybrid cuckoo search algorithm and earthworm algorithm	MATLAB
[66]	Heuristics and metaheuristics	Hybrid min-conflict local search algorithm and grey wolf optimization	MATLAB
[67]	Heuristics and metaheuristics	Bee colony	Netlogo
[68]	Heuristics and metaheuristics	Dragonfly algorithm and genetic algorithm	-
[69]	Heuristics and metaheuristics	Improved version of butterfly optimization algorithm	MATLAB
[70]	Heuristics and metaheuristics	Reinforcement learning	Python
[71]	Heuristics and metaheuristics	Reinforcement learning	-
[72]	Heuristics and metaheuristics	Reinforcement learning	-
[73]	Heuristics and metaheuristics	Reinforcement learning	Python

Reference	Category	Technique	Simulation Platform
[74]	Heuristics and metaheuristics	Reinforcement learning	MATLAB
[75]	Heuristics and metaheuristics	Reinforcement learning	Julia
[76]	Heuristics and metaheuristics	Reinforcement learning	MATLAB/Python
[77]	Other techniques	Modified Flower Pollination Algorithm and mixed-integer linear programming	MATLAB
[78]	Other techniques	Convolution neural network	MATLAB
[79]	Other techniques	Reinforcement learning and neural networks	MATLAB
[80]	Other techniques	Stochastic programming and robust programming	GAMS
[81]	Other techniques	Particle swarm optimization and two-point estimate method	MATLAB
[82]	Other techniques	Particle swarm optimization and sequential quadratic programming	MATLAB
[83]	Other techniques	Multi-objective genetic algorithm and multi-objective genetic programming	Python
[84]	Other techniques	Neural networks and model predictive control	MATLAB
[85]	Other techniques	Reinforcement learning and fuzzy reasoning	MATLAB
[86]	Other techniques	Fuzzy	MATLAB
[87]	Other techniques	Fuzzy	MATLAB
[88]	Other techniques	Fuzzy	MATLAB

 Table 5. Cont.

#### 3.1. Traditional Techniques (Mathematical Optimization)

In this article, the traditional optimization techniques for HEMS applications refer to techniques that use mathematical optimization based on exact algorithms. Hence, in this category it is considered that the solution to the problem is obtained using commercial solvers.

## 3.1.1. Linear Programming

Linear programming is an optimization technique where the objective function is a linear function and constraints are given by linear functions. Variables assume continuous values. It is considered the simplest way to represent mathematical programming problems. However, it may not be the most efficient way to represent complex real-world problems, mostly based on non-linear processes.

A comprehensive description of the theory of linear and integer programming may be found in [89].

Pure linear programming problems make it difficult and computationally costly to model more complex management systems such as HEMS, so it is not often used in the design of these systems, and this is one of the reasons why few authors in recent years adopted this approach. However, a sample of the employment of this method for HEMS may be found in [19], where the authors propose a HEMS that considers a PV-battery system and thermostatically controlled loads using linear programming to optimize the energy consumption of the loads. The objective is the minimization of operational costs.

The findings of selected papers associated with linear programming are shown in Table 6.

Reference	Findings
[19]	Comparison between a scheme without HEMS and with HEMS, proving the expected effectiveness of HEMS.

Table 6. Findings of selected papers associated with linear programming.

# 3.1.2. Mixed-Integer Linear Programming

Mixed-integer linear programming (MILP) refers to optimization techniques where the objective function is given by a linear function and subject to linear constraints but includes mixed variables, continuous and discrete variables. These problems allow for greater modeling power as it is possible to consider binary variables that help to represent real-world processes more effectively. The fact that it is based on a linear problem guarantees optimality conditions, i.e., it is guaranteed to obtain a global optimum. Many research works address the formulation of the HEMS problem using MILP. Most of the HEMS consider the load in a segmented matter (multiple appliances or thermal/electrical loads), as well as the energy price information.

A HEMS is proposed for a home with electric and thermal loads, and with the consideration of electric vehicles (EV) integration in [20].

MILP was used in [21] where a HEMS is presented to participate in the day-ahead electricity market through demand–response strategies. All loads, including thermal, are considered for demand response.

A price-sensitive HEMS is proposed by authors of [22], having as components a PV system, energy storage, and controllable loads.

Some authors design HEMS with a particular interest in managing energy storage systems (ESS). This is the case of the authors of [23].

Multiple sources of energy in the same residence are also object of study. For instance, the authors of [24] propose a HEMS for a house incorporating multi-energy resources, namely, electricity, thermal energy, renewables, energy storage, and natural gas.

A new two-level optimization algorithm for the energy management of residential appliances within a smart home is proposed in [25]. The system includes interruptible, uninterruptible, thermostatically controlled, and non-schedulable loads, as well as the charging/discharging strategies of EVs and ESS in the presence of distributed energy resources (DER).

Other studies considered more technical aspects of the residence energy management. An example may be found in [26], where the trading with technical influences of utility requirements is considered: Volt-Watt and Volt-Var functions.

The findings of selected papers associated with mixed-integer linear programming are shown in Table 7.

Reference	Findings			
[20]	The results of the proposed HEMS are compared with other energy management systems, showing the effectiveness of the proposed model, through case studies that allow reducing energy costs in both summer and winter.			
[21]	The results are compared when demand response is considered and when it is not. They demonstrate that the strategy presented with demand response is superior.			
[22]	The proposed methodology allows to reduce house costs by 53% and reduce peak-to-average ratio (PAR) by around 70%.			
[23]	Several energy-storage scenarios are considered, showing the benefits of considering a battery to store energy.			
[24]	The results show significant cost saving while maintaining user comfort.			
[25]	The results show a reduction in the electricity costs and an increase in power factor according to user's preference.			

Table 7. Findings of selected papers associated with mixed-integer linear programming.



The number of selected papers associated with a given objective for mixed-integer linear programming is shown in Figure 4.

**Figure 4.** Number of selected papers associated with a given objective for mixed-integer linear programming.

## 3.1.3. Non-Linear Programming

Non-linear programming, different from linear programming and MILP, is an optimization technique where the objective function and/or the constraints are given by non-linear functions. It represents real-world problems in a more real manner, but its main disadvantage is that it does not guarantee an optimal solution; that is, instead of a global optimal, a local optimal can be obtained. Additionally, the computation time tends to increase. For a comprehensive description of the non-linear programming theory, please refer to [90].

As is the case of linear programming, pure non-linear programming problems are not common in HEMS. However, the authors of [27] propose an energy management system for a residence composed solely of renewable sources, wind and PV, and batteries. The objective is to minimize the use of energy from the grid and maximize the sale of energy to the grid from renewables.

The findings of selected papers associated with non-linear programming are shown in Table 8.

Table 8. Findings of selected papers associated with non-linear programming.

Reference	Findings			
[27]	The results obtained by non-linear programming are compared with GA, showing the superiority of the former.			

3.1.4. Mixed-Integer Non-Linear Programming

Mixed-integer non-linear programming (MINLP) is an optimization technique where the objective function and/or the constraints are given by non-linear functions. Like MILP problems, they include mixed variables. Being non-linear does not guarantee obtaining a global optimum and they can be very difficult to be solved.

MINLP has been successfully applied to HEMS. For instance, the authors of [28] propose a HEMS with integrated renewables and energy storage.

A HEMS model for a smart home is proposed in [29] considering different capacities of energy storage and PV subsidy policies on the scheduling of household appliances.

The findings of selected papers associated with mixed-integer non-linear programming are shown in Table 9.

Reference	Findings			
[28]	Three operating scenarios are considered: normal, economic, and smart. The smart scenario manages to significantly reduce the daily energy cost, without greatly detracting from the other objectives.			
[29]	Comparison of single-objective and multi-objective optimization is considered. In the simulations a reduction of 61% and 71% is verified for peak load and electricity costs in the single-objective optimization.			

Table 9. Findings of selected papers associated with mixed-integer non-linear programming.

The number of selected papers associated with a given objective for mixed-integer non-linear programming is shown in Figure 5.



**Figure 5.** Number of selected papers associated with a given objective for mixed-integer non-linear programming.

# 3.1.5. Dynamic Programming

In dynamic programming (DP), the problem as a whole is divided into several subproblems, i.e., a solution to the original problem is obtained through simpler problems. The optimal solutions of the sub-problems are stored in time, which allows previous calculations to help in the next stages. Using this strategy, the complexity of the problem tends to decrease [91]. For the fundamentals for the theory of DP, please consult [92].

DP has been applied in HEMS. For instance, the authors of [30] proposed a HEMS for homes consisting of PV systems, batteries, and controllable loads. To solve the problem, the Differential DP (DDP) technique is presented.

A novel state-space approximate DP (SS-ADP) approach to quickly solve a HEMS problem can be found in [31].

The findings of selected papers associated with dynamic programming are shown in Table 10.

Table 10. Findings of selected papers associated with dynamic programming.

Reference	Findings			
[30]	The DDP results are compared with the results of the exact DP, approximate DP, and MINLP. DDP proves to be computationally faster than the mentioned techniques.			
[31]	Similar solutions are obtained by the proposed SS-ADP and DP, since the solutions of the SS-ADP approach are in the 0.8% of the DP solutions. In addition, savings in computational time of at least 20% are verified.			



The number of selected papers associated with a given objective for dynamic programming is shown in Figure 6.

Figure 6. Number of selected papers associated with a given objective for dynamic programming.

## 3.1.6. Stochastic Programming

Stochastic programming is an optimization technique where the objective function and constraints include uncertainty in parameters and variables. The optimal value of the objective function is given by the expected value of the objective function. Stochastic programming is considered when there is knowledge about the probability distribution function. Further, these problems can be based on two-state or multi-state problems, being called two-stage stochastic programming problems or multi-stage stochastic programming problems. A set of realizations for the parameters that involve uncertainty can be considered, being designated by scenarios. Usually, each scenario is associated with a probability. A detailed description of the fundamentals of stochastic programming may be found in [93].

Due to its characteristics, stochastic programming has been applied in HEMS, not only as the single method used but also associated with other methods to evaluate and deal with the uncertainty. To deal with the uncertainty, two situations in particular are explored with more frequency: dealing with electricity market price, especially when the system also injects energy in the grid, and dealing with multiple sources of renewable energy, in which the generation depends on factors that could suffer considerable variation, such as wind speed and solar radiation.

For instance, stochastic programming is employed by the authors of [32] for the optimal bidding strategy for autonomous residential energy management systems. This system allows the management of the production and consumption of the house and the participation in the local market environment through the technique of stochastic programming using intervals from the scenarios that represent the uncertainty of the market price of local electricity and PV production. The problem is formulated as a two-stage stochastic programming problem. The first stage is related to the day-ahead market and the second stage is related to real-time.

The approach of multiple-stage framework stochastic programming is explored by many authors. The optimal energy management and sizing of renewable energy for a home and a microgrid are proposed in [33], in which the problem is formulated in a two-stage approach.

In [34], a HEMS is presented for a house composed of the PV system and battery and loads. The problem is formulated as a two-stage stochastic programming problem as well. The first stage determines the optimal day-ahead energy procurement and the scheduling of shiftable appliances. The second stage is related to the real-time operation, namely the charging and discharging process of the battery.

In [35], a HEMS with PV systems, EVs, ESS, and thermal and electric loads is presented. The problem is formulated as a two-stage stochastic programming problem. The first stage defines the quantities of energy to be sold and purchased from the grid, and the second stage defines the decisions about customer's convenience and DER operation, namely temperature and charging and discharging rates of EVs and energy storage.

In the same fashion of multi-resource of energy as presented above, a HEMS is presented for a home that includes wind turbines, diesel generators, and EVs in [36].

The authors of [37] propose a HEMS with demand response, renewable resources, and battery storage.

In [38], a HEMS is presented, considering electrical and thermal loads.

The optimization problem is also formulated as two-stage stochastic programming in [39] for a HEMS including wind micro-turbine, battery, EV, and electric and thermal loads. The first stage is related to trading energy with the day-ahead local market, and the second stage is related to trading energy in real-time.

Another stochastic model of a HEM system is proposed by authors of [40]. The model optimizes the customer's cost in different demand response programs.

In [41], a HEMS is developed for a house consisting of different electric appliances and with generation from PV systems and battery storage.

The findings of selected papers associated with stochastic programming are shown in Table 11.

Reference	Findings			
[32]	It is demonstrated that the optimal offering model is more robust than the non-optimal offering model.			
[33]	The results of stochastic programming are compared with the results of deterministic programming, showing the superiority of the former.			
[34]	Stochastic programming is beneficial for using imperfect forecasts.			
[35]	The proposed strategy allows for a reduction in electricity costs.			
[36]	The results show that operating all capacity resources minimizes the operational costs.			
[37]	The operation is analyzed for two cases, with HEMS, and without HEMS. The use of a stochastic approach improved results.			
[38]	The proposed methodology allows obtaining better results in less time.			
[39]	Participation of smart home in a time-of-use pricing scheme leads to increased profit.			
[40]	The deterministic approach is compared with the stochastic approach, showing the superiority of the stochastic approach.			
[41]	The case studies presented on different prices allows to observe the effectiveness of the proposed model.			

Table 11. Findings of selected papers associated with stochastic programming.

The number of selected papers associated with a given objective for stochastic programming is shown in Figure 7.



Figure 7. Number of selected papers associated with a given objective for stochastic programming.

In robust programming, like stochastic programming, uncertainty is considered in the parameters and variables. There is no knowledge about the probability distribution function of parameters that involve uncertainty. Instead, intervals of values for the parameters are considered. This often leads to exaggerated assumptions about the parameters, i.e., it is considered the worst case. Consequently, it is not considered a consensual technique. Further details may be found in [94,95].

Despite the above disadvantages, robust programming has been applied to HEMS. For instance, the authors of [42] propose a HEMS considering a hierarchical control, having a central controller and local controllers. The central controller optimizes the schedule of the non-thermal loads. The local controllers respond to the real-time variations to obtain thermal comfort. A data-driven distributional robust optimization is proposed to guarantee solution robustness against the worst probability distribution of multiple uncertainties.

A flexible-constrained energy management model is proposed for smart-home-equipped PV systems and energy storage in [43].

In [44], a risk-based robust decision-making framework for smart residential buildings, considering electric and thermal appliances and plug-in hybrid vehicles is proposed.

The findings of selected papers associated with robust programming are shown in Table 12.

	Table 12. Findings	s of selected	papers associated	with robust	programming.
--	--------------------	---------------	-------------------	-------------	--------------

Reference	Findings		
[42]	The results show that the proposed approach can reduce the electricity cost in comparison with other techniques.		
[43]	The deterministic approach is compared with the robust approach. Additionally, a sensitivity analysis is performed to assess the robustness of the problem.		
[44]	The robustness of the problem and the computational efficiency are verified by the proposed approach.		

The number of selected papers associated with a given objective for robust programming is shown in Figure 8.



Figure 8. Number of selected papers associated with a given objective for robust programming.

#### 3.2. Model Predictive Control

The model predictive control (MPC) is an advanced method of control based on a receding horizon principle, aimed at determining the best course of action while meeting the requirements. In [96,97], detailed information on the theory of the MPC may be found.

The application of MPC in HEMS has developed significantly in recent years. For instance, in [45] a HEMS for a residential building with a PV system, ESS, thermal and electric loads, and EVs is proposed. The MPC problem considered a prediction horizon of four hours for every 5 min.

The authors of [46] propose a HEMS for a smart home focusing on the energy balance between the three phases to control both active and reactive power. Several case studies are considered, assuming a prediction horizon of 24 h, a control horizon of 24 h, and a simulation horizon of 48 h.

A comprehensive approach of a mixed-integer quadratic-programming MPC scheme based on the thermal building model and the building energy management system is employed by authors of [47].

A HEMS is developed by employing an MPC framework in [48] and implemented using a Branch-and-Bound algorithm. The authors discuss the selection of different parameters, such as time-step to employ, predict, and control horizons and the effect of the weather on the system performance.

A predictive HEMS for a residential building with the integration of a plug-in EV (PEV), a PV array, and a heat pump is developed by [49]. A stochastic MPC strategy is applied.

The findings of selected papers associated with model predictive control are shown in Table 13.

Reference	Findings			
[45]	Various configurations, namely, stand-alone and centralized configurations of MPC are formulated and compared with other traditional rule-based control. The MPC configurations suggest energy cost savings.			
[46]	The proposed methodology manages to balance the three phases.			
[47]	The proposed controller offers a solution to reformulate the discrete stochastic constraints to avoid the exponential growth of the scenario tree experienced by existing controllers. The proposed controller's efficiency is shown by benchmarking and comparing it with the multistage stochastic programming algorithm in the context of HEMS.			
[48]	A comparison is made between the economic performance of the proposed approach with a real photovoltaic battery system existing in a residence equipped with several IoT devices, with savings larger than 30% both on sunny and cloudy days.			
[49]	The MPC ensures that home load demand, PEV battery charging requirements, and household thermal comfort conditions are met.			

Table 13. Findings of selected papers associated with model predictive control.

The number of selected papers associated with a given objective for model predictive control is shown in Figure 9.



Figure 9. Number of selected papers associated with a given objective for model predictive control.

# Forecasting in HEMS

A smart household, as previously mentioned, is composed of a variety of energy systems. To allow the HEMS to make better decisions, knowing the future values—forecasting—of, for example, electric loads, electricity generation and storage state of charge, is an important asset for the improvement of HEMS efficiency.

The use of forecasting models is mandatory in model-based predictive control (MBPC). The reason is that MBPC uses predictive models, that should output the modelled variable's forecasts for each step ahead within the Prediction Horizon (PH) considered, i.e., provide multi-step-ahead forecasting. This type of forecast can be achieved in a direct mode, by having several one-step-ahead forecasting models, each providing the prediction of each step ahead within the prediction horizon. An alternative is to use a recursive version. In this case, only one model is necessary, and for each step within the PH, the inputs change, eventually employing predictions obtained in previous steps.

In this sub-section, we describe some of the articles in which the forecasting is a considerable component in the HEMS system.

In [51], a real-time forecasting is developed considering renewable generation, and the HEMS updates the inputs of scheduling system before each optimization calculation.

In [34], forecasting is employed in an integrated HEMS framework where it is assessed together with monitoring, scheduling, and coordination, focusing on the renewable energy generation and storage of the residence.

The authors of [98] develop power predictions using GA-ANN for a day-ahead forecast in a short-term fashion.

In [99], the authors used a Muti-Objective Genetic Algorithm (MOGA) framework to design a multi-step ahead recursive forecast model (a Radial Basis Function ANN) for the power demand in a residential HEMS. In [100], the same residence was used as case study to propose an ensemble forecasting approach. The ensemble of models is easily obtained by the MOGA approach and has been shown to obtain more accurate forecasts than the single solution. Although MOGA is a metaheuristic it is only used for model design, and not specifically for HEMS.

The authors of [84] use ANN to forecast the renewable energy sources (wind and solar) of a net zero energy building. In [101], the authors explore forecasting techniques in a HEMS from a prosumer perspective.

In [102], using the MOGA formulation, the authors develop a short-term forecasting of the photovoltaic solar power, to be used in a HEMS.

Forecasting is also very important in residential energy communities. In [103], the authors review a variety of forecasting methods to be employed in the energy communities control context. In [104], forecasting is explored considering the load demand, wind power, and electricity price.

## 3.3. Heuristics and Metaheuristics

Heuristics stand for strategies using readily accessible information to control problemsolving processes in man and machine, to obtain good enough results in an admissible length of time. Detailed information about heuristic search strategies may be found in [105,106]. A metaheuristic guides a subordinate heuristic using concepts derived from artificial intelligence, biological, mathematical, natural, and physical sciences to improve their performance. Information about metaheuristics theory may be found in [107,108]. Consequently, one of the differences between heuristics and metaheuristics is that heuristics are problem dependent; on the contrary, metaheuristics are problem independent.

Currently, heuristics and metaheuristics are the most used techniques in HEMS, taking advantage of the computational efficiency they often provide. For instance, a domestic microgrid is presented and an energy management system developed in [50]. The authors present four metaheuristics: grey wolf optimization (GWO), binary particle swarm optimization (PSO) (BPSWO), genetic algorithm (GA), and wind-driven optimization. Furthermore, they developed several combinations of these metaheuristics.

The authors of [51] propose an energy management system based on real-time electricity scheduling for a house that includes renewable, wind energy, and PV and ESS. GA is applied to obtain the solution of the problem.

A HEMS is presented in [52] that must guarantee the house's resilience, i.e., to be able to be self-sufficient when there is a failure in the network. To solve the problem, a new metaheuristic called Natural Aggregation Algorithm (NAA) is presented.

In [53] a HEMS is proposed for the optimal management of appliances in a home. To find the solution to the problem, four heuristics are presented: bat algorithm, GWO, moth flam optimization, and Harris hawks optimization.

A HEMS is presented in [54]. To obtain a solution to the problem, four metaheuristics are considered: integrated multi-objective antlion optimization, multi-objective PSO, the second version of the non-dominated sorting GA, and the basic antlion optimizer algorithm.

A HEMS for energy management at the level of electricity and heat is developed in [55]. To solve the problem, a fusion between the Harmony Search Algorithm technique and the PSO is presented.

The optimal planning of various energy sources, namely, fuel cell-based micro-combined heat and power (CHP), batteries, and EVs is proposed in [56]. For this, the real coded GA technique is presented.

A HEMS for a consumer in the presence of an energy storage and PV generation is developed by authors of [57]. To solve the problem, a hybrid technique between GWO and GA is presented, named Hybrid Grey Wolf GA (HGWGA).

In [58], a HEMS is proposed for a home with demand-responsive applications, PV systems, and ESS. The Polar Bear Optimization (PBO) technique is presented.

A scheduling algorithm for the energy management of a house consisting of PV and wind systems, batteries, diesel generators, and responsive loads is proposed in [59]. To solve the problem the PSO is used.

In [60], a HEMS is proposed for a smart home. To solve the problem, GWO is proposed. The authors of [61] propose a HEMS for a household considering DER and EVs. To solve the problem, a version of the GA, the improved GA, is presented.

A HEMS is proposed in [62], for a home with renewables, wind and PV, ESS, and electric and thermal loads. To solve the problem, GA is used.

In [63], a smart grid scenario is developed with a novel restricted and multi-restricted scheduling method for the residential customers. The optimization problem is developed under the time-of-use (TOU) pricing scheme. To optimize the formulated problem, GWO is utilized.

The problem of optimal scheduling of home appliances in HEM systems is formulated as a constrained, multi-objective optimization problem with integer decision variables and a powerful variant of PSO, named as enhanced leader (EL) PSO in [64]. In the proposed multi-objective formulation, the effect of weight factor on optimal electricity bill of the home and optimal comfort of the consumers is meticulously investigated.

In [65], an optimization problem for HEMS using a hybrid approach based on cuckoo search algorithm and earthworm algorithm is developed. However, there is a problem in such HEMS, that is, an uncertain behavior of the user that can lead to forced start or stop of an appliance, deteriorating the purpose of scheduling of appliances. To solve this issue, coordination among appliances for rescheduling is incorporated in HEMS using game theory.

The min-conflict local search algorithm (MCA) is hybridized with the GWO for the power scheduling problem in smart home in [66]. The proposed method is called GWO-MCA. MCA is utilized as a new operator of GWO to improve its exploitation capability in addressing constraint satisfaction problems, particularly scheduling problems.

The authors of [67] propose a computational intelligence model for the Internet of Things applications by applying the concept of swarm intelligence into connected devices. The bee colony approach is used.

The HEMS performance improvement by using a dragonfly algorithm and GA heuristicbased approach is evaluated in [68].

In [69], a new optimal method is developed for HEMS based on the Internet of Things. The problem is solved using an improved version of the butterfly optimization algorithm (BOA).

The findings of selected papers associated with heuristics and metaheuristics excluding reinforcement learning are shown in Table 14.

**Table 14.** Findings of selected papers associated with heuristics and metaheuristics excluding reinforcement learning.

Reference	Findings				
[50]	The hybrid versions guarantee lower costs than the mentioned heuristics.				
[51]	The results of GA-based ESS management and traditional GA are compared. GA-based ESS management is found superior to traditional GA.				
[52]	NAA is compared with Differential Evolutionary and PSO. NAA shows better results both in terms of obtaining results and computing time.				
[53]	The Harris hawks optimization technique obtains the best results compared to the other techniques.				
[54]	The integrated multi-objective antlion optimization can reduce costs by more than 80%.				
[55]	The fusion between the two techniques demonstrates superiority compared to the separate techniques.				
[56]	Optimum operation of energy sources is achieved by considering various types of tariffs.				
[57]	The HGWGA technique outperforms GWO and GA.				
[58]	PBO is compared with GWO and GA, showing superiority in reducing electricity and PAR costs. Still, PBO proves to be faster in terms of computing.				
[59]	Three scenarios are considered, ranging from an unmanaged system to the proposed planning, showing the effectiveness of the proposed model.				
[60]	GWO is compared with other techniques. The results show that GWO allows for better results.				
[61]	The results show that improved GA gives better results than GA.				
[62]	The results show that the proposed methodology allows for better results.				
[63]	GWO is compared with the PSO algorithm to show its effectiveness.				
[64]	The results indicate the superiority of ELPSO over basic PSO, artificial bee colony, backtracking search algorithm, gravitational search algorithm, and dragonfly algorithm.				
[65]	The results show a reduction in electricity costs of 50%.				
[66]	The hybrid technique outperforms other techniques.				
[67]	The proposed approach shows a reduction in electricity costs.				
[68]	Dragonfly shows better results than GA in electricity cost reduction.				
[69]	The improved version of BOA is compared with PSO and BOA showing the effectiveness of the proposed approach.				

The number of selected papers associated with a given objective for heuristics and metaheuristics excluding reinforcement learning is shown in Figure 10.





#### Reinforcement Learning (RL)

RL is one of the techniques in the heuristics and metaheuristics category that has grown the most in recent years. In this sense, a specific section on this technique is created.

A HEMS is developed in [70] using the reinforcement learning (RL) technique. Through the Q-learning algorithm, it is possible to obtain an appliance operation planning.

The authors of [71] propose a HEMS focusing on demand response. To solve the problem the RL technique is presented.

The energy management for a residential building is presented in [72]. The building has PV panels, EVs, and micro-CHP. To solve the problem, the RL technique is used.

The authors of [73] developed a HEMS for a home with electric and thermal loads, PV systems, energy storage, and EVs. To solve the problem, the RL is employed, more precisely deep Q-learning and double deep Q-learning.

The authors of [74] provide a steady price prediction model based on artificial neural networks. In cooperation with forecasted future prices, multi-agent RL is adopted to make optimal decisions for different home appliances in a decentralized manner.

In [75], based on the living habits of the residents, dependency modes for house energy resources are proposed and are integrated into the RL algorithms. Through the case studies, it is verified that the proposed method can schedule the appliances properly to satisfy the established dependency modes.

In [76], a data-driven approach that leverages RL to manage the optimal energy consumption of a smart home with a rooftop solar photovoltaic system, ESS, and smart home appliances is developed. The same authors propose a hierarchical deep RL (DRL) method for the scheduling of energy consumption of smart home appliances and DER including an ESS and an EV [109].

The findings of selected papers associated with reinforcement learning are shown in Table 15.

Reference	Findings				
[70]	The RL technique is compared with Least Slack Time (LST) scheduling, obtaining better results.				
[71]	To show the effectiveness of the proposed model, the RL results are compared with the MILP results, showing the superiority of the RL.				
[72]	The results demonstrate that RL promotes good planning, allowing the purchase of energy and gas to be carried out in periods of low prices.				
[73]	Double deep Q-learning is compared with PSO, showing better results.				
[74]	The results show that the strategy with demand response is better than the strategy without demand response.				
[75]	The difference between the achieved result by the proposed approach and the optimal solution is relatively small. However, the computational time for DRL is smaller than optimization.				
[76]	The results show a reduction of 14% in the electricity costs.				
[109]	The results show that the proposed methodology allows for better results.				

Table 15. Findings of selected papers associated with reinforcement learning.

The number of selected papers associated with a given objective for reinforcement learning is shown in Figure 11.



Figure 11. Number of selected papers associated with a given objective for reinforcement learning.

## 3.4. Other Techniques

This category includes all other techniques that do not fit into the categories presented or that have hybrid versions of the techniques described above.

Scheduling and optimal planning of appliances are approached by different authors. In [77], a HEMS for the optimal planning of appliances, energy resources, and electrothermal storage is presented. For this, the combination between the Modified Flower Pollination Algorithm and the MILP is presented.

The authors of [78] provide a new residential energy management system based on the convolution neural network (CNN) including a PV array environment. The CNN is used in the estimation of the non-linear relationship between the residence PV array power and meteorological datasets. The residential energy management system has three main stages for the energy management such as forecasting, scheduling, and real functioning. A short-term forecasting strategy has been performed in the forecasting stage based on the PV power and the residential load. A coordinated scheduling has been utilized for minimizing the cost function.

A novel framework for HEMS based on RL and employing neural networks in achieving efficient home-based demand response is addressed in [79].

A HEMS for a smart home having a PV system and battery storage is proposed in [80]. To solve the problem, a hybrid technique between stochastic programming and robust optimization is presented.

The authors of [81] propose a HEMS with a focus on demand response. To solve the problem, a hybrid technique between the PSO and the two-point estimate method is used. The technique is shown to guarantee shorter computation times without reducing the accuracy of the results.

The energy management for a house consisting of ESS, fuel cell and electrical and thermal loads is proposed in [82]. To solve the problem, a hybrid technique between PSO and sequential quadratic programming is presented.

A multi-objective optimization of offline strategies for HEMS is addressed in [83]. Two approaches are compared, namely the common timetable-based and the proposed approach based on decision trees. The timetable-based strategy is optimized using multi-objective GA and the tree-based using multi-objective genetic programming. The results of the latter show a reduction in costs of up to 17%.

A novel technique for managing the energy in a zero-energy building using a combination of neural networks and MPC is proposed in [84].

A combination of RL and fuzzy reasoning is employed in [85] for an effective energy management system with demand response. RL is considered as a model-free control strategy which learns from the interaction with its environment by performing actions and evaluating the results.

The authors of [86] propose the design and implementation of a fuzzy control system that processes environmental data to recommend minimum energy consumption values for a residential building. This system follows the forward chaining Mamdani approach and uses decision tree linearization for rule generation.

In [87], the implementation of a HEMS based on a fuzzy logic controller is addressed. The proposed HEMS manages the energy from the PV to supply home appliances in the grid-connected PV-battery system. Similarly, a new power management strategy based on fuzzy logical combined state machine control is developed in [88]. Its effectiveness is compared with various strategies such as DP, state machine control, and fuzzy logical control with simulation.

The findings of selected papers associated with other techniques are shown in Table 16.

Reference	Findings				
[77]	The hybrid technique allows to obtain superior results in less time. Additionally, it is compared with other metaheuristics and better results are achieved.				
[78]	The CNN is compared with other methods, proving its superiority.				
[79]	The results show the effectiveness of the data-driven HEMS framework, both for electricity costs and computational efficiency.				
[80]	Robust optimization is used in the day-ahead operation and stochastic programming for the real time operation. The results prove the effectiveness of the hybrid approach.				
[81]	The hybrid technique guarantees shorter computation times without reducing the accuracy of the results.				
[82]	The authors claim that this hybrid technique allows to incorporate the global search characteristics of PSO with the local search capabilities of sequential quadratic programming. The results demonstrate the effectiveness of the proposed model.				
[84]	The results show that the system is capable of satisfying loads without the need to import energy from the grid. On the contrary, the proposed approach allows injecting energy into the grid and having some economic return.				
[85]	The results show that the proposed scheduling approach can smooth the power consumption and reduce the electricity costs.				
[86]	The proposed approach improved the accuracy and made the computation faster.				
[87]	Experimental results show a fast processing time and a cost reduction of 10% when compared with a system without a fuzzy logic controller.				
[88]	The effectiveness of the proposed approach is compared with various strategies such as DP, state machine control, and fuzzy logical control with simulation.				

Table 16. Findings of selected papers associated with the other techniques category.



The number of selected papers associated with a given objective for other techniques is shown in Figure 12.



# 4. Computational Issues Considered in Home Energy Management Systems

When designing the HEMS, the decision maker must consider several factors, among which are optimal decision making and the associated computational issues. In the case of computational issues, one must consider the resources available at the level of processing the programs, the formulation of the problem, and the time window that allows new decisions to be made for the optimal HEMS operation. These issues arise because there is a need for timely decision making and because of the dimension of the problem that tends to grow into more complex problems.

# 4.1. Problem Size

The literature states that the increase in the problem dimension is related to several aspects, namely, with the parameters that model the considered time intervals, such as the resolution and the planning horizon, and with the number of devices considered [18].

However, other causes can increase the scale of HEMS problems. One of the causes for the increase in the dimension of problems in the field of engineering, especially in energy management in buildings, is the consideration of uncertainty. Uncertainty is an essential issue because buildings in the future, in addition to the inherent uncertainty related to load, will use renewable energy sources, and integrate EVs. Here, we are talking about the increase in the dimension of the problem due to uncertainty essentially when considering stochastic programming. When considering the representation of uncertainty through scenarios in a problem, a reasonable number of scenarios should be considered to represent the uncertainty more accurately. However, considering too many scenarios can lead to an increase in computing time incompatible with timely decision making [93]. To solve this problem, the literature has suggested scenario reduction techniques in [110,111]. Scenario reduction is an important branch of decision-making problems in problems involving uncertainty. The concept of scenario reduction aims to define subsets of the initial set of scenarios, i.e., to determine a set of representative scenarios that best represent the initial set of scenarios [110,111]. In the initial set, each scenario is associated with a probability. In the final set, the preserved scenarios assume new probabilities that were previously associated with the initial scenarios.

#### 4.2. Planning Horizon, Resolution, and Time Steps

From a computational point of view, in HEMS design it is important to present the concepts of planning horizon, resolution, and time steps. Planning horizon refers to the period of operation that is subject to treatment and optimization by the HEMS, generally in periods of more than 1 h to days. Resolution refers to the time interval considered between each decision-making period, usually in periods of minutes to hours. Time steps are the result of the ratio between the planning horizon and the resolution. For instance, a planning horizon of 12 h is considered in [48], with a resolution of 15 min, and 48 time steps. A planning horizon of 24 h is considered, together with a resolution of 15 min, making a total of 96 time steps in [22,24,42,52,55,72]. The scheduling horizon considered ranges from 5 a.m. of day D to 5 a.m. of day D + 1. Similarly, in [21,28,32,39,50,70,112], the time horizon is also 24 h, but with a resolution of 1 h and 24 time steps. Other works propose the use of a 30 min resolution for a 24 h time horizon, which is equivalent to 48 time steps [23,27,51,61]. There are cases where optimization is performed for shorter resolution periods, e.g., 5 min for a time horizon of 24 h and consequently 288 time steps are involved [20,25,34,59]. Some authors present cases where optimization is performed for several resolution values. This is the case of [53] which compares the results for a time horizon of 24 h, but for two different resolutions: 1 h and 12 min. The work with the lowest resolution among the selected papers is [60], where the resolution is 1 min for a time horizon of 24 h and a time steps value of 1440. The authors of [19,37,46,73,113] propose planning for a longer period, in the order of days to years.

The planning horizon, the resolution, and time steps used in the selected papers are shown in Table 17.

<b>Planning Horizon</b>	Resolution	Time Steps	References
12 h	15 min	48	[48]
24 h	1 min	1440	[60,87]
24 h	5 min	288	[20,25,34,59]
24 h	12 min	120	[53,87]
24 h	15 min	96	[22,24,26,42,52,55,72]
24 h	30 min	48	[23,26,27,41,51,61,64]
24 h	1 h	24	[21,28,30–33,36,39,40,43,49,50,53,54,56, 57,62,68,70,71,76,77,79,80,82,85,109,112]
Days to years			[19,37,46,73,75,113]

Table 17. Planning horizon, the resolution, and time steps used in the selected papers.

Table 17 shows that, as expected, the 24 h planning horizon and the 1 h resolution are dominant.

#### 5. Technical Issues Considered in Home Energy Management Systems

The decision maker must also consider some important technical aspects when designing a HEMS. Namely, it must consider whether there is a need to consider uncertainty in the formulation of the problem, whether there is more than one objective to be considered, and whether it is necessary to formulate the problem as a multi-objective problem. Additionally, demand-side management strategies and the associated tariffs and incentives should also be discussed.

## 5.1. Uncertainty

The energy management of a house is a complex problem that requires extra attention to the components that make up the house system. Particularly, nowadays houses are increasingly based on energy production systems based on renewable energy resources due to environmental concerns. In this sense, the production of energy from renewable sources implies greater flexibility in energy management systems, as the forecast of its production is never perfect. Additionally, the load, i.e., the power consumption and the electricity market prices also have random and difficult to predict behaviors. In recent years, yet another source of uncertainty has begun to hover over electrical systems and HEMS, the integration of EVs, many of them as bidirectional devices, with grid-to-vehicle and vehicle-to-grid capability.

Proper representation of procedures through mathematical descriptions cannot always be formalized using deterministic formulations. In the real world, it is expected that situations occur where the representation of parameters through their average values is not favorable. The literature presents solutions for non-deterministic environments to include the treatment of uncertainty. According to [114], decision making considering the treatment of uncertainty can be divided into three types, according to the environment: decision making with random parameters in which the probability distribution function is known; decision making with parameters where there is no information about the probability distribution function; and decision making based on fuzzy sets theory, with uncertainty in parameters' fuzzy numbers and in constraint fuzzy sets.

To reduce the effects of uncertainty in a house, some techniques are presented in the literature to solve this issue that can technically and economically affect the operation of energy management systems in houses.

In environments where the probability distribution is known, the use of stochastic programming techniques is suggested, namely two-stage stochastic programming and multi-stage stochastic programming. Thus, in [32] the uncertainty management using two-stage stochastic programming is presented. In the day-ahead stage, the proposed residential energy management system considers uncertainty for electricity price and PV generation modeled by interval-based scenarios. In the real-time stage, the uncertainty is modeled by scenarios. There are 4 scenarios for day-ahead stage and they have the same probability, while for real-time stage 10 scenarios are considered, with variable probability.

The authors of [33] addressed the uncertainty related to energy production, with load consumption and with electricity prices. For each hour of the day, 10 scenarios are generated for each of the uncertainty sources.

The authors of [34,36] addressed the uncertainty related to PV power output and load, and wind power and load, respectively. For each uncertainty resource 6 scenarios are generated, making a total of 36 scenarios. The probabilities of the scenarios are variable.

Some works consider the need to reduce the number of scenarios when there is a high degree of uncertainty. This is the case of [35] where 1500 scenarios are generated to represent PV power behavior, ambient temperature, and load behavior. Due to the high number of scenarios, the number of 1500 scenarios is reduced to 15 new scenarios by applying the backward method scenario reduction technique, to which new probabilities are assigned.

The authors of [38] also used scenario reduction when considering uncertainty of PV power. The scenario generation is based on a seven-step distribution model for a Gaussian distribution, resulting in a total of 4096 scenarios. This value is reduced to 10 scenarios using the backward reduction method proposed by [110]. The proposed technique is compared with Monte Carlo Simulation (MCS) and two-point method. However, to obtain a cost like the proposed technique, the MCS must increase the scenarios from 50,000 to 500,000, but that obviously implies a very large increase in computation time. Additionally, considering the two-point method that generates 48 scenarios, there is no guarantee of representing the uncertainty in the best way.

In [39], the uncertainty is associated with mobility pattern of EVs and wind power. The scenarios are generated by an Autoregressive Moving Average (ARMA) and reduced by 10 scenarios by K-means clustering technique.

The uncertainty associated with PV power, wind power, load, and electricity price are addressed in [37]. A maximum of 450 scenarios is considered to represent uncertainty, as, according to the authors, computing time can increase significantly.

The authors of [40] consider uncertainties of EVs availability and small-scale renewable energy generation, PV power, and wind power. Parameter realizations with uncertainty are modeled through roulette wheel mechanism (RWM).

In [41], the uncertainty in PV power and energy demand is considered. In total, 10 scenarios of sources of uncertainty are considered.

In environments where the probability distribution is not known, the use of robust programming techniques is necessary. Thus, in [43] the uncertainty management using robust programming is presented. Uncertainty is related to electricity market prices. In this work, it is stated that the use of stochastic programming is not the ideal technique to model the uncertainty of market prices, because the probability density functions may not correctly represent the price behavior with the integration of renewable sources. Furthermore, it is said that the best way to access market price uncertainty without using historical data is by considering the worst case, obtained through robust optimization.

A robust optimization approach is considered to deal with the uncertainty on energy supply, based on wind power and the uncertainty of thermal and electrical loads by the authors of [44]. It is stated that robust optimization approaches can be better suited for assuming that the parameters are within some limits, which is more general than assuming a known distribution of parameters. Furthermore, it is found that historical data can be important for updating the limits, and that robust optimization is more suitable for problems that involve risk, as they assume that the scenarios are the worst-case scenario within the parameter limits.

Fuzzy decision-making environment is the last group. In this environment, according to [114] two types of uncertainty exist, including ambiguity and vagueness. Ambiguity refers to the situation where the choice among multiple alternatives is undetermined, while vagueness refers to the case in which sharp and precise boundaries for some domains of interest are not delineated.

#### 5.2. Multi-Objective Problems

Usually, the management of a home is presented in the literature only to respond to a single objective, or formulated only for the electrical part, or for the thermal part. However, in many cases, researchers consider multiple objectives to be achieved simultaneously, as was possible to see in the summary of many of the works presented in the previous sections. In the literature, multiple objectives are presented simultaneously, as minimization of energy costs and user comfort [24], minimization of energy costs, PAR and user comfort [50,53,54], minimization of energy cost and emissions [77], minimization of energy costs and PAR [50,57], and minimization of energy costs and battery degradation [59]. However, these papers do not present the formulation in the conventional form of multi-objective problems, that is, problems with multi-objective formulations. Thus, in a multi-objective problem we are in the presence of more than one objective to be maximized or minimized following specific rules. The most used techniques in the specialized literature on HEMS when there is a multi-objective problem are weighted sum and the representation technique Pareto Front. However, outside the specialized literature on HEMS, other techniques such as bounded objective and physical programming are used for multi-objective problems [18].

#### 5.2.1. Weighted Sum

In weighted sum, the formulation of the problem consists of the definition of an objective function composed of two or more objective functions, and to each of these objective functions a scalar is added that aims to define the weight that each will have in the final value of the resulting objective function. Generally, the objective functions that make up the scalar objective function are functions that concur with distinct or conflicting objectives. The sum of the weights of each of the objective functions typically assumes the unity value. It is considered the simplest way to formulate multi-objective problems due to its simplicity, but the value of the objective must be linearly scalable [18].

For instance, a multi-objective problem is formulated by the authors of [70], aiming to minimize electricity costs and minimize the discomfort wait cost. A coefficient is used to balance the cost of electricity with the discomfort wait cost.

Similarly, the HEMS proposed in [20] is formulated as a multi-objective problem, employing as minimization objectives the energy consumption expense and comfort deviation. A total of 11 groups of combinations between the coefficients that model the weight of each objective are considered. It is verified that this formulation allows flexibility to adjust the results according to user preferences.

A problem formulated to answer multiple objectives is presented in [28], namely, energy costs, user's convenience, consecutive waiting time, and PAR. At this stage, the goals all have the same weight as they do not have associated coefficients. However, given the results obtained, it is verified that it is in the user's interest to associate a higher weight to some of the goals. Thus, in a second phase, coefficients are associated to each objective, resulting in a multi-objective problem using the weighted sum technique. It is concluded that this technique allows easier access to specific requirements of residents.

The authors of [23] formulate the model as a multi-objective problem that is converted into a single objective problem by a linear combination of two objective functions. The two conflicting goals are the reduction in energy costs and the reduction in peak load demand.

In [22], the HEMS is formulated as a multi-objective problem using weight coefficients for two distinct objectives, as are user's comfort and the total operating cost which are negatively correlated.

An original problem is proposed in [46] with three different objectives, namely, selfconsumption, user comfort, and grid support. To formulate the problem, three coefficients are considered that model the weight of each individual objective. The authors state that this work presents an innovation, as the three weights can be permanently adjusted and used in the next simulation step.

The authors of [60] claim that the formulation of the scheduling problem as a single objective may lead to some issues in balancing the power demand and improving the user comfort level due to the larger effort on minimizing energy cost in detriment of the other objectives. Therefore, a multi-objective formulation is proposed, having four objectives to minimize, namely the cost of electricity, PAR, capacity power limit rate, and waiting time rate.

In [44], the energy management of the residential buildings is formulated as a multiobjective problem, considering two objectives, namely operating and CO<sub>2</sub> emissions costs. Using the weighted sum technique, two weights are employed to control the impact of the system operation and the total emissions on the total cost of the system; thus making it possible to make a trade-off between the environmental impact and the economic impact.

# 5.2.2. Pareto Front

In multi-objective problems it is important to introduce a fundamental concept which is the Pareto Front concept. Pareto Front refers to a trade-off curve where all the optimal values of a multi-objective problem are based [115]. This curve has the function of offering the decision maker a set of optimal solutions on which it then decides which one is closest to its ambitions [114,115]. Thus, in a Pareto Front, with two conflicting objectives, one can only get a higher value for one of the objective problem using the Pareto front technique. The two objectives in dispute are the minimization of energy consumption and the minimization of energy costs. To show the effectiveness of the formulation, the Pareto Front results are compared with the results obtained by the multi-objective problem formulated with weighted sum. The results of Pareto Front are superior to those of weighted sum, achieving, in many cases, a reduction in cost of about 25% while an increase in the use of renewable sources is achieved.

The objectives associated with multi-objective problems obtained from the list of selected papers are shown in Table 18.

Reference	Objectives	Weighted Sum	Pareto Front
[70]	Minimization of electricity costs and of discomfort wait cost	1	
[20]	Minimization of energy consumption expense and comfort deviation	1	
[28]	Minimization of energy costs, maximization of user's convenience, minimization of consecutive waiting time, and minimization of PAR	1	
[23]	Minimization of energy costs and reduction in peak load demand	1	
[22]	Maximize user's comfort and minimization of operating costs	1	
[46]	Maximize self-consumption, user comfort, and grid support	1	
[60]	Minimization of electricity costs, PAR, capacity power limit rate, and waiting time rate	1	
[62]	Minimization of energy costs and of energy consumption		1
[44]	Minimization of operating cost and CO <sub>2</sub> emissions	1	

Table 18. Objectives associated with multi-objective problems from the list of the selected papers.

Table 18 shows that in most of the selected papers the formulation is based on multiobjective problems using the weighted sum technique.

# 5.3. User Behavior

When developing HEMS, user behavior must also be taken into account. Several issues can be addressed within the scope of user behavior, namely: the number of occupants in the house, which can result in a significant increase in energy consumption; the availability of users to participate in demand response strategies; and the behavior of users with electric vehicles, which is a trend that is here to stay. For instance, in [20] six types of uncertain user behaviors are considered, which are separately modeled through energy consumption deviation. These six types of behaviors form the set of uncertain behaviors that allow predicting the needs of users. A scheduling algorithm based on human-appliances interaction in smart homes is proposed in [70]. This strategy of considering humanappliance interaction proves to have good results. In [45], user behavior is introduced in the model of a HEMS, through historical data extracted from a smart meter. It is also stated that constraints may arise in data acquisition in buildings due to privacy issues when considering user behavior. In [21], an electric vehicle charging management system is integrated into a HEMS, taking into account the behavior of electric vehicle owners. To model the stochastic behavior of electric vehicle holders, the uncertainty formulation takes into account the arrival and departure periods of electric vehicles.

#### 5.4. Demand Response and Pricing

Electric systems were characterized a few decades ago, in the previous paradigm, as being essentially centralized and with little or no user action. Currently, there is a paradigm shift towards a network where the user can assume some status to be able to contribute to decisions related to the electrical system. In this sense, one of the clearest examples of this paradigm shift are the incentives that aim to encourage the user to reduce or change energy consumption. Here, we enter the field of demand response, which is defined as changing the profile of energy use by users based on dynamic electricity prices or other incentives to improve the efficiency and reliability of the power system [7]. Demand

response programs take the form of time varying pricing or rebates because of changing the demand profile compared to the forecasted base value [116]. In this sense, demand response programs are commonly classified as price-based or incentive-based programs [6,7,116]. According to [6], both categories aim to minimize the energy bill for the users, with the following differences: in price-based programs, demand is adjusted to network prices according to user preferences, namely, TOU price, real-time price (RTP), critical peak price (CPP), and inclining block rate (IBR); while incentive-based programs can be divided into voluntary–direct load control (DLC) or emergency demand response program (EDRP); mandatory–capacity market program (CAP) or interruptible/curtailable services (I/C); and market clearing–demand bidding (DB) or ancillary services (A/S).

TOU is a widely used pricing scheme, characterized by being made up of two or three different price periods over the course of a day. In the case of two prices, the first refers to the price for off-peak periods and the second to peak periods [7]. The three-price case consists of low-peak, mid-peak, and peak periods [7,117]. In many cases, the TOU is characterized by having an associated seasonality, and it is usual for the hours of application of these prices to vary, for example, in summer and winter. Additionally, the timetable for these prices varies with the day of the week, with weekdays having one cost and Saturday and Sunday other costs. This is the case for companies operating in Portugal.

For instance, in [112] the TOU is applied as a pricing scheme together with RTP or for fixed prices. The TOU guarantees the greatest reduction in operating costs and the fixed price the highest cost. Furthermore, the TOU guarantees the greatest exchange of energy with the grid and the RTP the lowest value.

A typical Moroccan household is managed with a HEMS considering the TOU in [27]. The step-rate tariff is considered, which together with the TOU are the two most used price schemes in Morocco.

The TOU is one of the pricing schemes proposed in a HEMS to manage a house considering demand response in [45]. The TOU is compared with hourly pricing and five minutes pricing. The result of one year of operation shows that the TOU allows cost savings of up to 26%, a value higher than that obtained by the hourly pricing. However, the results demonstrate that the real-time five minutes pricing allows cost savings of up to 42%. These results are said to show the effectiveness of the TOU and RTP.

CPP is a pricing scheme very similar to the TOU, considering two distinct periods, off-peak and peak periods, just like the TOU, with the difference of considering two periods, one of extremely high demand values and another with relatively low demand [7,117]. The existence of extremely high demand periods causes prices to follow this behavior, which leads to CPP prices being higher than the TOU. CPP is a very specific case of price, as it refers to days where less usual demand profiles are seen, characterized by very large peaks.

RTP is a pricing scheme where prices vary hourly, and there can be very large differences between the price of one hour and another hour. Two types of RTP are generally considered, namely, day-ahead pricing and hourly pricing [7]. In the case of day-ahead pricing, the prices for each hour of day D are defined on day D - 1. This price is closely related to the hourly prices that are defined in the day-ahead electricity market and that reflect the marginal price, resulting from the crossing of the energy purchase and sale curves. The hourly price is defined by prices revealed every hour [7]. Day-ahead pricing is the most suitable for consumers, as they know a priori the price of each hour, allowing for timely planning and with a greater degree of certainty [7,117].

For instance, the RTP scheme is subscribed by a user of a home in Illinois, US, in [51]. The next 2 h RTP signal of each period is available for the consumer through a smart meter. The developed HEMS has the capability of bi-directional operation, being able to sell energy to the network at half the price of the RTP scheme.

The RTP scheme is used in [54] to measure the energy costs of a home. The RTP scheme is compared with CPP scheme. The problem is formulated as a multi-objective problem that analyzes the effects of each of the tariffs on each of the objectives, which are the electricity cost and the PAR. The results show that the electricity cost is reduced by 75%

and 80% when using RTP and CPP, respectively. Regarding PAR, it is reduced by 35% and 30% using RTP and CPP schemes with respect to the unscheduled case.

Similarly, RTP and CPP schemes are used in [57] in a HEMS. The electricity cost has been reduced by up to 43% when only the RTP scheme is considered. By incorporating the CPP scheme, the cost of electricity was reduced by up to 63% and the PAR reduced by up to 38%.

Inclining Block Rate (IBR) is a pricing scheme where the price increases with increasing energy consumption. Thus, an increase in energy consumption that exceeds the value defined for an hour, a day, or a month result in an increase in the price to be charged [7]. This pricing scheme forces users to be more careful in how they consume energy over a given period. The IBR scheme is used in conjunction with other price-based demand response schemes in a HEMS in [41]. The other pricing schemes evaluated are RTP, CPP, and TOU. The results show that the RTP scheme is the best method applied, as it simulates real market conditions. It is said that with the TOU, as there are no penalties, it is only effective for cases with low consumption values. Furthermore, it is claimed that the IBR together with the TOU scheme can solve the problem of over-consumption using the price signals.

#### 6. From HEMS to Energy Communities

Although the focus of this work is not energy communities and microgrids, we briefly present here the developments that have been made in these areas in recent years, as those are a very hot topic of research and application

An energy management system for a residential microgrid with a micro-CHP unit, photovoltaic generation, hybrid vehicles, and ESS is proposed in [112]. The objective is to minimize the energy costs associated with the purchase of energy from the grid, and with the cost of natural gas. The transaction is compared for various pricing scenarios, namely, TOU, RTP, or for fixed prices.

The authors of [113] propose the comparison between a management system for a house and a community. Systems must manage energy from thermal and electrical points of view. The problem is formulated as a multi-objective system, where the objectives are the minimization of operating costs and emissions. The system is applied for a planning horizon of one year and one week. It is concluded that the Community Management System is better than the Management System for a house, both economically and environmentally. It is stated that neither the Home Management System nor the Community Management System is economically viable at this time, unless the size of the ESS is reduced, and the thermal energy capacity increased.

In [103], a review of microgrid control and optimization in terms of objectives, constraints, and optimization techniques is presented. The components of the microgrid, the advantages for the grid and for the final consumers are also presented.

Another review is presented in [118] regarding power generation systems for microgrids for residential applications. Greater attention is paid to DC microgrids. The technical differences and strategies used in energy management systems in microgrids are presented.

A hierarchical energy management system is proposed for a set of building microgrids that together form an energy community in [119]. Each building has a building energy management system that optimizes the operation considering local information and price information. Next, at the highest level, the community energy management system is responsible for managing the operation between all building microgrids and then with the main network. This sequence of decision making allows the minimization of the operation costs of the problem formulated as a MILP problem. A system for planning the operation of a microgrid is presented in [120]. The microgrid consists of several participants with renewable production and EVs. The technique used to find the optimal solution is the multiagent RL. Multi-agent RL is compared to single-agent RL, demonstrating the superiority of the former. The authors of [104] addressed an energy management system for a commercial microgrid that includes renewable production from wind and the inclusion of EVs. The paper's biggest contribution is the consideration of electricity price uncertainty in an MPC.

Furthermore, the risk is considered through the Conditional Value at Risk (CVaR) measure. The objective is the minimization of operating costs. It is verified that EVs have an effective contribution to the microgrid energy balance.

In [121], an energy management system for microgrids consisting of a set of buildings, ESS, EV, and renewable energy sources is presented. The PSO is applied to find a solution to the problem that aims to minimize costs. The results demonstrate a significant reduction in costs.

A fuzzy logic-based energy management for microgrids connected to the grid is developed by the authors of [122]. The system includes renewable sources and energy storage. The goal is to minimize power fluctuations in the network, while maintaining the battery state of charge within predefined limits.

Similarly, in [123], a fuzzy logic controller is applied in the EMS of microgrids by considering dramatic behavior of renewable energy resources while maintaining battery state within secure limits. A comparison with the time-based constant and variable charge/discharge control of battery demonstrates, in simulation, the effectiveness of the proposed fuzzy controller in a residential AC microgrid.

An energy management system is presented for a hybrid residential microgrid consisting of a diesel generator, wind turbine, PV system, and battery energy storage in [124]. The objectives are to minimize costs, reduce emissions, and increase the penetration of renewable sources. The problem is solved using PSO. The results demonstrate that there can be a reduction in emissions of more than 35% in the optimal configuration compared to the scenario where diesel generators are responsible for satisfying all demand.

A risk-averse energy management system for large-scale industrial building microgrids is presented in [125]. The objectives are to minimize operating costs and minimize greenhouse gas emissions. To solve the problem, a hybrid technique between the flower pollination algorithm and MILP is applied. It is noted that this hybrid version is applied because the techniques compensate for the limitations of each one, namely the disadvantage of obtaining an optimal location with the flower pollination algorithm and the size of the problem for the MILP. The results show that the proposed hybrid technique outperforms other techniques such as GA and simulated annealing.

The authors of [126] propose an energy management system of a building microgrid with the objective of minimizing energy costs, considering thermal comfort. The system uses the MPC to define the optimal operating points. The results allow for the reduction in operating costs.

A model for the optimal integration of building microgrids based on renewable sources, EV, and batteries in the electricity market is proposed in [127], namely, in the day-ahead market and in the regulated market. The objective is to maximize the profit of building microgrids through participation in the electricity market. The problem is formulated as a MILP problem. The results demonstrate the increase in profit, the reduction in renewable energy curtailment, and decrease in power in peak hours compared to the case where there is no optimization.

Stochastic programming is applied in an energy management system for grid-connected building microgrids considering uncertainty of PV power and the stochastic behavior of EVs in [128]. The system based on stochastic programming is compared with deterministic programming, showing that the former guarantees the minimization of electricity costs.

In [129], an energy management system for a residential microgrid with PV generation and battery energy storage is presented. The aim is to minimize electricity costs. PSO is used to solve the problem. The PSO when compared with other techniques allows better results to be obtained.

# 7. Conclusions

This review article provided a comprehensive systematic review of published works considering HEMS techniques, technical and computational issues. The results of the systematic review resulted in the division of techniques used in HEMS systems into four

categories: traditional techniques, MPC, heuristics and metaheuristics, and other techniques. The authors conclude this document by presenting the discussion of the following points identified through the development of this work, and that should be highlighted:

- IEEE Access (IEEE) is the journal with the biggest number of documents published within the review criteria, followed by *Energies* (MDPI) and Applied Energy (Elsevier). Within the considered period, 2020 was the year when most documents were published.
- The reviews published between 2018 and 2021 focused mainly on broader concepts and enabling technologies presentation, HEMS architecture, benefits for the energy grid, and specific components of HEMS or methods, as reinforced learning.
- There is a variety of documents being published on the topic of HEMS. However, when addressing specifically the HEMS techniques, many studies were excluded from the review due to the lack of information on the applied method itself.
- Although not being a specific search term, most of the articles have as an objective the reduction in the economic costs related to residence power consumption.
- The statistical analysis of the data showed that the most used technique is the category of heuristics and metaheuristics, followed by traditional techniques. In fact, this was an expected conclusion since with the development of systems that emulate the behavior of nature and the development of Artificial Intelligence, it has seen an exponential growth in recent years. Thus, it is expected that in the future these systems will continue to play an important role in HEMS design.
- It is also important to highlight the traditional techniques, based on exact algorithms and solved by commercial solvers, which guarantee a global optimum when formulated as linear problems. This is an advantage of traditional techniques and should not be overlooked. With the evolution of computational power and the reduction in computing time, these techniques will continue to play an important role in the future of HEMS.
- The MPC will also be an important player as it can contribute to limiting the uncertainty that characterizes HEMS.
- MATLAB is the simulation platform most used for the development of selected works, but some works also used GAMS and Python, among other punctual uses of other platforms.
- In computational issues, it is verified that most of the papers address the issue of operation planning using HEMS using a 24 h planning horizon and a 1 h resolution. It is expected that in the future there will be a need for optimization systems that plan for values of less than one hour to obtain a more reliable optimization.
- Some documents, despite clearly presenting the HEMS technique being used, did not present in detail the simulation setup used, which may make the method complicated to replicate. The application of the same method for the same case study/data set may have different results according to the assumed simulation setup, such as planning horizon, resolution, and time steps. We advise the clear presentation of this setup in future publications.
- Uncertainty is also a very important factor to consider in HEMS systems, especially when it integrates renewable energy resources and their forecast.
- Most HEMS ideally should have a multi-objective approach, once the social component related to the occupants' preferences is as important as the energy savings or other objectives intrinsically related to grid or system benefits.
- Overall, the works published on HEMS are very focused on the adaptation of the adopted framework to the case study in which it is applied, and ideally there should be more studies focusing on the assessment of generalization of the methods for other datasets (residential buildings with different occupation patterns, equipment usage, among others). The good performance of the HEMS generalization assessment contributes as a basis to the commercialization of these solutions.

The performance assessment metrics of a HEMS—how good is the HEMS for a determined case study—should vary according to the objectives of the system, considering the compromise between accuracy, computational resources, computational time, among others.

Although this review followed all the rules presented in Section 2, this work has some limitations. From the outset, the fact of considering selected papers from the year 2018 forward on the Web of Science platform. This decision was made as the optimization in the HEMS area has grown rapidly and we aimed to present a systematic and recent review that could provide an overview of what has been done in recent years. As a result, some important papers in the area may have been missed. However, we presented throughout the text, and, in Table 1, the review papers that were previously undertaken in this area and that include papers earlier than 2018.

We found that it is very difficult to compare the results obtained with different technologies, and even with different approaches using the same technology. This is due to the complexity of the problem, the multiplicity of the different factors intervening in a HEMS, and the lack of public databases in this topic that can be used by different researchers to assess their different technologies. In this sense, in [130] the reader can find several data related to a group of houses located in the South of Portugal, with HEMS, renewable energy, and energy storage.

As this work was focused on HEMS techniques, nothing was mentioned regarding the necessary infrastructure to implement the HEMS, quite often obtained by IOT techniques. Additionally, to implement HEMS techniques efficiently, one should be able to detect which appliances are working, preferably using Non-Invasive Load Monitoring (NILM) methods. These are important topics for an overall view of HEMS.

Another limitation is the absence of a complete review of microgrids and energy communities. We only briefly present the link between HEMS, energy communities, and microgrids. In the future, from our point of view, it is expected that management systems for energy communities will become the rule rather than the exception as it has been until now. This will happen because with the opening of the market and with greater decision-making power of the prosumers, they will aggregate to start managing their energy on a larger scale and perhaps enter the market in the form of clean energy aggregators. When a large community of buildings is considered, its management involves typically the fast processing of a large quantity of data, precluding, this way, the use of big data platforms.

Author Contributions: Conceptualization, M.G.R. and A.R.; methodology, I.G., K.B. and A.R.; software, I.G., K.B.; validation, M.G.R. and A.R.; formal analysis, I.G., K.B., M.G.R. and A.R.; investigation, I.G., K.B., M.G.R. and A.R.; resources, M.G.R. and A.R.; data curation, I.G., K.B., M.G.R. and A.R.; writing—original draft preparation, I.G., K.B. and A.R.; writing—review and editing, I.G., K.B., M.G.R. and A.R.; visualization, I.G., K.B., M.G.R. and A.R.; supervision, M.G.R. and A.R.; project administration, M.G.R. and A.R.; funding acquisition, M.G.R. and A.R. All authors have read and agreed to the published version of the manuscript.

**Funding:** This research was funded by Operational Program Portugal 2020 and Operational Program CRESC Algarve 2020, grant numbers 39578/2018 and 72581/2020. Antonio Ruano also acknowledges the support of Fundação para a Ciência e Tecnologia, grant UID/EMS/50022/2020, through IDMEC under LAETA.

Informed Consent Statement: Not applicable.

**Data Availability Statement:** Data supporting reported results can be found at https://csi.ualg.pt/nilmforihem (accessed on 31 November 2021).

**Conflicts of Interest:** The authors declare no conflict of interest.

# Nomenclature

ANN	Artificial neural network
A/S	Ancillary services
ARMA	Autoregressive Moving Average
BOA	Butterfly optimization algorithm
BPSWO	Binary particle swarm optimization
CAP	Capacity market program
CHP	Combined heat and power
CNN	Convolution neural network
CPP	Critical peak pricing
DB	Demand bidding
DER	Distributed energy resources
DDP	Differential dynamic programming
DLC	Direct load control
DP	Dynamic programming
DRL	Deep reinforcement learning
EDRP	Emergency demand response program
ELPSO	Enhanced leader particle swarm optimization
ESS	Energy storage system
EV	Electric vehicle
GA	Genetic algorithm
GWO	Grey Wolf Optimization
HEMS	Home energy management system
IBR	Inclining Block Rate
I/C	Interruptible/curtailable services
HGWGA	Hybrid Grey Wolf GA
LST	Least Slack Time
MCA	Min-conflict local search algorithm
MOGA	Multi-objective genetic algorithm
MILP	Mixed-integer linear programming
MINLP	Mixed-integer non-linear programming
NAA	Natural Aggregation Algorithm
NILM	Non-intrusive load monitoring
MPC	Model predictive control
PAR	Peak-to-average ratio
PBO	Polar bear optimization
PEV	Plug-in EV
PSO	Particle swarm optimization
PV	Photovoltaic
RWM	Roulette wheel mechanism
RL	Reinforcement learning
RTP	Real-time price
TOU	Time-of-use
SS-ADP	State-space approximate DP

# Appendix A

The number of citations and location of selected review articles with at least five citations is shown in Table A1.

Table A1. Number of citations and location of selected review articles with at least 5 citations.

Reference	Citations	Location
[5]	15	India
[6]	38	Portugal
[7]	38	Malaysia

Reference	Citations	Location
[9]	41	USA
[11]	17	Qatar
[12]	132	United Arab Emirates
[13]	77	Malaysia
[14]	9	USA
[16]	11	India
[17]	49	Spain

Table A1. Cont.

The number of citations was taken from the Web of Science as of 23 March 2022.

The number of citations and location of selected articles with at least five citations is shown in Table A2.

Table A2. Number of citations and location of selected articles with at least 5 citations.

Reference	Citations	Location
[19]	21	Iraq
[20]	27	China
[23]	59	India
[24]	20	China
[28]	6	South Korea
[29]	16	China
[32]	30	Spain
[33]	18	Russia
[35]	16	USA
[36]	43	Qatar
[37]	23	Turkey
[38]	31	China
[39]	21	Spain
[40]	112	Italy
[41]	23	Iran
[44]	45	Iran
[45]	37	USA
[46]	7	Austria
[47]	47	Austria
[49]	20	Denmark
[50]	24	Pakistan
[51]	33	China
[54]	19	Iran
[55]	5	China
[56]	8	South Korea
[57]	20	Pakistan

Reference	Citations	Location
[58]	5	Pakistan
[59]	7	Algeria
[60]	13	United Arab Emirates
[61]	8	China
[62]	7	Mexico
[63]	24	Ethiopia
[64]	6	Iran
[65]	10	Pakistan
[66]	13	Malaysia
[67]	7	South Korea
[68]	13	Pakistan
[69]	25	China
[72]	10	Iran
[73]	13	China
[74]	79	South Korea
[76]	26	South Korea
[79]	49	China
[80]	21	Iran
[81]	70	China
[82]	13	China
[83]	15	Slovenia
[84]	18	Egypt
[85]	26	England
[86]	9	Greece
[88]	8	South Korea

Table A2. Cont.

The number of citations was taken from the Web of Science as of 23 March 2022. The location is based on the available information about the first author.

#### References

- 1. Moyer, J.D.; Hughes, B.B. ICTs: Do They Contribute to Increased Carbon Emissions? *Technol. Forecast. Soc. Chang.* 2012, 79, 919–931. [CrossRef]
- Ponce, P.; Polasko, K.; Molina, A. End User Perceptions toward Smart Grid Technology: Acceptance, Adoption, Risks, and Trust. *Renew. Sustain. Energy Rev.* 2016, 60, 587–598. [CrossRef]
- 3. Eiselt, H.A.; Sandblom, C.-L. *Linear Programming and Its Applications*; Springer Science & Business Media: New York, NY, USA, 2007; ISBN 3540736719.
- 4. Kitchenham, B.; Charters, S. Guidelines for Performing Systematic Literature Reviews in Software Engineering; Keele University: Keele, UK, 2007.
- Mahapatra, B.; Nayyar, A. Home Energy Management System (HEMS): Concept, Architecture, Infrastructure, Challenges and Energy Management Schemes. *Energy Syst.* 2019, 1–27. [CrossRef]
- 6. Leitao, J.; Gil, P.; Ribeiro, B.; Cardoso, A. A Survey on Home Energy Management. IEEE Access 2020, 8, 5699–5722. [CrossRef]
- 7. Makhadmeh, S.N.; Khader, A.T.; Al-Betar, M.A.; Naim, S.; Abasi, A.K.; Alyasseri, Z.A.A. Optimization Methods for Power Scheduling Problems in Smart Home: Survey. *Renew. Sustain. Energy Rev.* **2019**, *115*, 109362. [CrossRef]
- 8. Badar, A.Q.H.; Anvari-Moghaddam, A. Smart Home Energy Management System—A Review. *Adv. Build. Energy Res.* 2020, 16, 118–143. [CrossRef]
- Mason, K.; Grijalva, S. A Review of Reinforcement Learning for Autonomous Building Energy Management. *Comput. Electr. Eng.* 2019, 78, 300–312. [CrossRef]

- 10. Yu, L.; Qin, S.; Zhang, M.; Shen, C.; Jiang, T.; Guan, X. A Review of Deep Reinforcement Learning for Smart Building Energy Management. *IEEE Internet Things J.* 2021, *8*, 15. [CrossRef]
- Zafar, U.; Bayhan, S.; Sanfilippo, A. Home Energy Management System Concepts, Configurations, and Technologies for the Smart Grid. IEEE Access 2020, 8, 119271–119286. [CrossRef]
- 12. Shareef, H.; Ahmed, M.S.; Mohamed, A.; al Hassan, E. Review on Home Energy Management System Considering Demand Responses, Smart Technologies, and Intelligent Controllers. *IEEE Access* **2018**, *6*, 24498–24509. [CrossRef]
- 13. Hannan, M.A.; Faisal, M.; Ker, P.J.; Mun, L.H.; Parvin, K.; Mahlia, T.M.I.; Blaabjerg, F. A Review of Internet of Energy Based Building Energy Management Systems: Issues and Recommendations. *IEEE Access* **2018**, *6*, 38997–39014. [CrossRef]
- 14. McIlvennie, C.; Sanguinetti, A.; Pritoni, M. Of Impacts, Agents, and Functions: An Interdisciplinary Meta-Review of Smart Home Energy Management Systems Research. *Energy Res. Soc. Sci.* **2020**, *68*, 101555. [CrossRef]
- Parvin, K.; Lipu, M.S.H.; Hannan, M.A.; Abdullah, M.A.; Jern, K.P.; Begum, R.A.; Mansur, M.; Muttaqi, K.M.; Mahlia, T.M.I.; Dong, Z.Y. Intelligent Controllers and Optimization Algorithms for Building Energy Management Towards Achieving Sustainable Development: Challenges and Prospects. *IEEE Access* 2021, *9*, 41577–41602. [CrossRef]
- 16. Rajendhar, P.; Jeyaraj, B.E. Application of DR and Co-Simulation Approach for Renewable Integrated HEMS: A Review. *IET Gener. Transm. Distrib.* **2019**, *13*, 3501–3512. [CrossRef]
- Mariano-Hernández, D.; Hernández-Callejo, L.; Zorita-Lamadrid, A.; Duque-Pérez, O.; García, F.S. A Review of Strategies for Building Energy Management System: Model Predictive Control, Demand Side Management, Optimization, and Fault Detect & Diagnosis. J. Build. Eng. 2021, 33, 101692.
- Beaudin, M.; Zareipour, H. Home Energy Management Systems: A Review of Modelling and Complexity. *Renew. Sustain. Energy Rev.* 2015, 45, 318–335. [CrossRef]
- 19. al Essa, M.J.M. Home Energy Management of Thermostatically Controlled Loads and Photovoltaic-Battery Systems. *Energy* **2019**, 176, 742–752. [CrossRef]
- 20. Lu, Q.; Lü, S.; Leng, Y.; Zhang, Z. Optimal Household Energy Management Based on Smart Residential Energy Hub Considering Uncertain Behaviors. *Energy* 2020, 195, 117052. [CrossRef]
- 21. Baek, K.; Ko, W.; Kim, J. Optimal Scheduling of Distributed Energy Resources in Residential Building under the Demand Response Commitment Contract. *Energies* **2019**, *12*, 2810. [CrossRef]
- 22. Lyu, J.; Ye, T.; Xu, M.; Ma, G.; Wang, Y.; Li, M. Price-Sensitive Home Energy Management Method Based on Pareto Optimisation. *Int. J. Sustain. Eng.* **2021**, *14*, 433–441. [CrossRef]
- 23. Lokeshgupta, B.; Sivasubramani, S. Multi-Objective Home Energy Management with Battery Energy Storage Systems. *Sustain. Cities Soc.* **2019**, *47*, 101458. [CrossRef]
- 24. Wu, X.; Li, Y.; Tan, Y.; Cao, Y.; Rehtanz, C. Optimal Energy Management for the Residential MES. *IET Gener. Transm. Distrib.* 2019, 13, 1786–1793. [CrossRef]
- 25. Zamanloo, S.; Abyaneh, H.A.; Nafisi, H.; Azizi, M. Optimal Two-Level Active and Reactive Energy Management of Residential Appliances in Smart Homes. *Sustain. Cities Soc.* **2021**, *71*, 102972. [CrossRef]
- Dao, V.T.; Ishii, H.; Takenobu, Y.; Yoshizawa, S.; Hayashi, Y. Home Energy Management Systems under Effects of Solar-Battery Smart Inverter Functions. *IEEJ Trans. Electr. Electron. Eng.* 2020, 15, 692–703. [CrossRef]
- Azaroual, M.; Ouassaid, M.; Maaroufi, M.; Author, C. An Optimal Energy Management of Grid-Connected Residential Photovoltaic-Wind-Battery System Under Step-Rate and Time-of-Use Tariffs. Int. J. Renew. Energy Res. 2020, 10, 1829–1843.
- 28. Dinh, H.T.; Kim, D. An Optimal Energy-Saving Home Energy Management Supporting User Comfort and Electricity Selling with Different Prices. *IEEE Access* 2021, *9*, 9235–9249. [CrossRef]
- 29. Lu, Q.; Zhang, Z.; Lü, S. Home Energy Management in Smart Households: Optimal Appliance Scheduling Model with Photovoltaic Energy Storage System. *Energy Rep.* 2020, *6*, 2450–2462. [CrossRef]
- Jeddi, B.; Mishra, Y.; Ledwich, G. Differential Dynamic Programming Based Home Energy Management Scheduler. *IEEE Trans.* Sustain. Energy 2020, 11, 1427–1437. [CrossRef]
- Zhao, Z.; Keerthisinghe, C. A Fast and Optimal Smart Home Energy Management System: State-Space Approximate Dynamic Programming. *IEEE Access* 2020, *8*, 184151–184159. [CrossRef]
- Shokri Gazafroudi, A.; Soares, J.; Fotouhi Ghazvini, M.A.; Pinto, T.; Vale, Z.; Corchado, J.M. Stochastic Interval-Based Optimal Offering Model for Residential Energy Management Systems by Household Owners. *Int. J. Electr. Power Energy Syst.* 2019, 105, 201–219. [CrossRef]
- Farrokhifar, M.; Aghdam, F.H.; Alahyari, A.; Monavari, A.; Safari, A. Optimal Energy Management and Sizing of Renewable Energy and Battery Systems in Residential Sectors via a Stochastic MILP Model. *Electr. Power Syst. Res.* 2020, 187, 106483. [CrossRef]
- 34. Zheng, Z.; Sun, Z.; Pan, J.; Luo, X. An Integrated Smart Home Energy Management Model Based on a Pyramid Taxonomy for Residential Houses with Photovoltaic-Battery Systems. *Appl. Energy* **2021**, *298*, 117159. [CrossRef]
- 35. Mohsenzadeh, A.; Pang, C. Two Stage Residential Energy Management under Distribution Locational Marginal Pricing. *Electr. Power Syst. Res.* **2018**, *154*, 361–372. [CrossRef]
- 36. Mehrjerdi, H.; Hemmati, R. Coordination of Vehicle-to-Home and Renewable Capacity Resources for Energy Management in Resilience and Self-Healing Building. *Renew. Energy* 2020, 146, 568–579. [CrossRef]

- 37. Merdanoğlu, H.; Yakıcı, E.; Doğan, O.T.; Duran, S.; Karatas, M. Finding Optimal Schedules in a Home Energy Management System. *Electr. Power Syst. Res.* 2020, 182, 106229. [CrossRef]
- Du, Z.; Wang, W.; Zhang, J.; Zhang, Y.; Xu, X.; Liu, J. A Data-Driven Home Energy Scheduling Strategy under the Uncertainty in Photovoltaic Generations. *IEEE Access* 2020, *8*, 54125–54134. [CrossRef]
- Gazafroudi, A.S.; Shafie-khah, M.; Heydarian-Forushani, E.; Hajizadeh, A.; Heidari, A.; Corchado, J.M.; Catalão, J.P.S. Two-Stage Stochastic Model for the Price-Based Domestic Energy Management Problem. *Int. J. Electr. Power Energy Syst.* 2019, 112, 404–416. [CrossRef]
- 40. Shafie-Khah, M.; Siano, P. A Stochastic Home Energy Management System Considering Satisfaction Cost and Response Fatigue. *IEEE Trans. Ind. Inform.* 2017, 14, 629–638. [CrossRef]
- 41. Javadi, M.S.; Gough, M.; Lotfi, M.; Nezhad, A.E.; Santos, S.F.; Catalão, J.P.S. Optimal Self-Scheduling of Home Energy Management System in the Presence of Photovoltaic Power Generation and Batteries. *Energy* **2020**, *210*, 118568. [CrossRef]
- 42. Saberi, H.; Zhang, C.; Dong, Z.Y. Data-Driven Distributionally Robust Hierarchical Coordination for Home Energy Management. *IEEE Trans. Smart Grid* **2021**, *12*, 4090–4101. [CrossRef]
- Bahramara, S. Robust Optimization of the Flexibility-Constrained Energy Management Problem for a Smart Home with Rooftop Photovoltaic and an Energy Storage. J. Energy Storage 2021, 36, 102358. [CrossRef]
- Golpîra, H.; Khan, S.A.R. A Multi-Objective Risk-Based Robust Optimization Approach to Energy Management in Smart Residential Buildings under Combined Demand and Supply Uncertainty. *Energy* 2019, 170, 1113–1129. [CrossRef]
- 45. Mirakhorli, A.; Dong, B. Market and Behavior Driven Predictive Energy Management for Residential Buildings. *Sustain. Cities Soc.* **2018**, *38*, 723–735. [CrossRef]
- Rao, B.V.; Kupzog, F.; Kozek, M. Phase Balancing Home Energy Management System Using Model Predictive Control. *Energies* 2018, 11, 3323. [CrossRef]
- Killian, M.; Zauner, M.; Kozek, M. Comprehensive Smart Home Energy Management System Using Mixed-Integer Quadratic-Programming. *Appl. Energy* 2018, 222, 662–672. [CrossRef]
- 48. Bot, K.; Laouali, I.; Ruano, A.; Ruano, M.d.G. Home Energy Management Systems with Branch-and-Bound Model-Based Predictive Control Techniques. *Energies* **2021**, *14*, 5852. [CrossRef]
- 49. Yousefi, M.; Hajizadeh, A.; Soltani, M.N.; Hredzak, B. Predictive Home Energy Management System With Photovoltaic Array, Heat Pump, and Plug-In Electric Vehicle. *IEEE Trans. Ind. Inform.* **2020**, *17*, 430–440. [CrossRef]
- 50. Iqbal, Z.; Javaid, N.; Iqbal, S.; Aslam, S.; Khan, Z.A.; Abdul, W.; Almogren, A.; Alamri, A. A Domestic Microgrid with Optimized Home Energy Management System. *Energies* **2018**, *11*, 1002. [CrossRef]
- 51. Li, S.; Yang, J.; Song, W.; Chen, A. A Real-Time Electricity Scheduling for Residential Home Energy Management. *IEEE Internet Things J.* **2019**, *6*, 2602–2611. [CrossRef]
- 52. Yang, Y.; Wang, S. Resilient Residential Energy Management with Vehicle-to-Home and Photovoltaic Uncertainty. *Int. J. Electr. Power Energy Syst.* **2021**, *132*, 107206. [CrossRef]
- 53. Mouassa, S.; Bouktir, T.; Jurado, F. Scheduling of Smart Home Appliances for Optimal Energy Management in Smart Grid Using Harris-Hawks Optimization Algorithm. *Optim. Eng.* **2021**, *22*, 1625–1652. [CrossRef]
- 54. Ramezani, M.; Bahmanyar, D.; Razmjooy, N. A New Optimal Energy Management Strategy Based on Improved Multi-Objective Antlion Optimization Algorithm: Applications in Smart Home. *SN Appl. Sci.* **2020**, *2*, 1–17. [CrossRef]
- Zhang, Z.; Wang, J.; Zhong, H.; Ma, H. Optimal Scheduling Model for Smart Home Energy Management System Based on the Fusion Algorithm of Harmony Search Algorithm and Particle Swarm Optimization Algorithm. *Sci. Technol. Built Environ.* 2020, 26, 42–51. [CrossRef]
- 56. Rafique, M.K.; Haider, Z.M.; Mehmood, K.K.; Uz Zaman, M.S.; Irfan, M.; Khan, S.U.; Kim, C.H. Optimal Scheduling of Hybrid Energy Resources for a Smart Home. *Energies* 2018, *11*, 3201. [CrossRef]
- Iqbal, M.M.; Sajjad, I.A.; Amin, S.; Haroon, S.S.; Liaqat, R.; Khan, M.F.N.; Waseem, M.; Shah, M.A. Optimal Scheduling of Residential Home Appliances by Considering Energy Storage and Stochastically Modelled Photovoltaics in a Grid Exchange Environment Using Hybrid Grey Wolf Genetic Algorithm Optimizer. *Appl. Sci.* 2019, *9*, 5226. [CrossRef]
- 58. Iqbal, M.M.; Zia, M.F.; Beddiar, K.; Benbouzid, M. Optimal Scheduling of Grid Transactive Home Demand Responsive Appliances Using Polar Bear Optimization Algorithm. *IEEE Access* 2020, *8*, 222285–222296. [CrossRef]
- Bouakkaz, A.; Haddad, S.; Andrés Martín-García, J.; Gil-Mena, A.J.; Jiménez-Castañeda, R. Optimal Scheduling of Household Appliances in Off-Grid Hybrid Energy System Using PSO Algorithm for Energy Saving. *Int. J. Renew. Energy Res.* 2019, 9, 427–436.
- Makhadmeh, S.N.; Al-Betar, M.A.; Alyasseri, Z.A.A.; Abasi, A.K.; Khader, A.T.; Damaševičius, R.; Mohammed, M.A.; Abdulkareem, K.H. Smart Home Battery for the Multi-Objective Power Scheduling Problem in a Smart Home Using Grey Wolf Optimizer. *Electronics* 2021, 10, 447. [CrossRef]
- 61. Kong, X.; Sun, B.; Kong, D.; Li, B. Home Energy Management Optimization Method Considering Potential Risk Cost. *Sustain. Cities Soc.* **2020**, *62*, 102378. [CrossRef]
- Gutierrez-Martinez, V.J.; Moreno-Bautista, C.A.; Lozano-Garcia, J.M.; Pizano-Martinez, A.; Zamora-Cardenas, E.A.; Gomez-Martinez, M.A. A Heuristic Home Electric Energy Management System Considering Renewable Energy Availability. *Energies* 2019, 12, 671. [CrossRef]

- 63. Molla, T.; Khan, B.; Moges, B.; Alhelou, H.H.; Zamani, R.; Siano, P. Integrated Optimization of Smart Home Appliances with Cost-Effective Energy Management System. *CSEE J. Power Energy Syst.* **2019**, *5*, 249–258. [CrossRef]
- 64. Jordehi, A.R. Enhanced Leader Particle Swarm Optimisation (ELPSO): A New Algorithm for Optimal Scheduling of Home Appliances in Demand Response Programs. *Artif. Intell. Rev.* **2020**, *53*, 2043–2073. [CrossRef]
- 65. Jamil, A.; Alghamdi, T.A.; Khan, Z.A.; Javaid, S.; Haseeb, A.; Wadud, Z.; Javaid, N. An Innovative Home Energy Management Model with Coordination among Appliances Using Game Theory. *Sustainability* **2019**, *11*, 6287. [CrossRef]
- Makhadmeh, S.N.; Khader, A.T.; Al-Betar, M.A.; Naim, S.; Abasi, A.K.; Alyasseri, Z.A.A. A Novel Hybrid Grey Wolf Optimizer with Min-Conflict Algorithm for Power Scheduling Problem in a Smart Home. *Swarm Evol. Comput.* 2021, 60, 100793. [CrossRef]
- 67. Bui, K.N.; Agbehadji, I.E.; Millham, R.; Camacho, D.; Jung, J.J. Distributed Artificial Bee Colony Approach for Connected Appliances in Smart Home Energy Management System. *Expert Syst.* **2020**, *37*, e12521. [CrossRef]
- 68. Hussain, I.; Ullah, M.; Ullah, I.; Bibi, A.; Naeem, M.; Singh, M.; Singh, D. Optimizing Energy Consumption in the Home Energy Management System via a Bio-Inspired Dragonfly Algorithm and the Genetic Algorithm. *Electronics* **2020**, *9*, 406. [CrossRef]
- 69. Wang, X.; Mao, X.; Khodaei, H. A Multi-Objective Home Energy Management System Based on Internet of Things and Optimization Algorithms. *J. Build. Eng.* **2021**, *33*, 101603. [CrossRef]
- Diyan, M.; Silva, B.N.; Han, K. A Multi-Objective Approach for Optimal Energy Management in Smart Home Using the Reinforcement Learning. Sensors 2020, 20, 3450. [CrossRef]
- Mathew, A.; Roy, A.; Mathew, J. Intelligent Residential Energy Management System Using Deep Reinforcement Learning. *IEEE Syst. J.* 2020, 14, 5362–5372. [CrossRef]
- Ahrarinouri, M.; Rastegar, M.; Seifi, A.R. Multiagent Reinforcement Learning for Energy Management in Residential Buildings. IEEE Trans. Ind. Inform. 2021, 17, 659–666. [CrossRef]
- 73. Liu, Y.; Zhang, D.; Gooi, H.B. Optimization Strategy Based on Deep Reinforcement Learning for Home Energy Management. *CSEE J. Power Energy Syst.* 2020, *6*, 572–582. [CrossRef]
- Lu, R.; Hong, S.H.; Yu, M. Demand Response for Home Energy Management Using Reinforcement Learning and Artificial Neural Network. *IEEE Trans. Smart Grid* 2019, 10, 6629–6639. [CrossRef]
- Si, C.; Tao, Y.; Qiu, J.; Lai, S.; Zhao, J. Deep Reinforcement Learning Based Home Energy Management System with Devices Operational Dependencies. Int. J. Mach. Learn. Cybern. 2021, 12, 1687–1703. [CrossRef]
- 76. Lee, S.; Choi, D.-H. Reinforcement Learning-Based Energy Management of Smart Home with Rooftop Solar Photovoltaic System, Energy Storage System, and Home Appliances. *Sensors* **2019**, *19*, 3937. [CrossRef] [PubMed]
- 77. Ali, S.; Malik, T.N.; Raza, A. Risk-Averse Home Energy Management System. IEEE Access 2020, 8, 91779–91798. [CrossRef]
- Chellaswamy, C.; Babu, R.G.; Vanathi, A. A Framework for Building Energy Management System with Residence Mounted Photovoltaic. In Proceedings of the Building Simulation, Bruges, Belgium, 1–3 September 2021; Tsinghua University Press: Beijing, China, 2021; Volume 14, pp. 1031–1046.
- 79. Xu, X.; Jia, Y.; Xu, Y.; Xu, Z.; Chai, S.; Lai, C.S. A Multi-Agent Reinforcement Learning-Based Data-Driven Method for Home Energy Management. *IEEE Trans. Smart Grid* 2020, *11*, 3201–3211. [CrossRef]
- Akbari-Dibavar, A.; Nojavan, S.; Mohammadi-Ivatloo, B.; Zare, K. Smart Home Energy Management Using Hybrid Robust-Stochastic Optimization. *Comput. Ind. Eng.* 2020, 143, 106425. [CrossRef]
- 81. Huang, Y.; Wang, L.; Guo, W.; Kang, Q.; Wu, Q. Chance Constrained Optimization in a Home Energy Management System. *IEEE Trans. Smart Grid* 2018, *9*, 252–260. [CrossRef]
- 82. Huang, Y.; Wang, W.; Hou, B. A Hybrid Algorithm for Mixed Integer Nonlinear Programming in Residential Energy Management. J. Clean. Prod. 2019, 226, 940–948. [CrossRef]
- Zupančič, J.; Filipič, B.; Gams, M. Genetic-Programming-Based Multi-Objective Optimization of Strategies for Home Energy-Management Systems. *Energy* 2020, 203, 117769. [CrossRef]
- Megahed, T.F.; Abdelkader, S.M.; Zakaria, A. Energy Management in Zero-Energy Building Using Neural Network Predictive Control. *IEEE Internet Things J.* 2019, 6, 5336–5344. [CrossRef]
- Alfaverh, F.; Denai, M.; Sun, Y. Demand Response Strategy Based on Reinforcement Learning and Fuzzy Reasoning for Home Energy Management. *IEEE Access* 2020, *8*, 39310–39321. [CrossRef]
- Kontogiannis, D.; Bargiotas, D.; Daskalopulu, A. Fuzzy Control System for Smart Energy Management in Residential Buildings Based on Environmental Data. *Energies* 2021, 14, 752. [CrossRef]
- Soetedjo, A.; Nakhoda, Y.I.; Saleh, C. Embedded Fuzzy Logic Controller and Wireless Communication for Home Energy Management Systems. *Electronics* 2018, 7, 189. [CrossRef]
- Ou, K.; Yuan, W.; Choi, M.; Kim, J.; Kim, Y. Development of New Energy Management Strategy for a Household Fuel Cell/Battery Hybrid System. Int. J. Energy Res. 2019, 43, 4686–4700. [CrossRef]
- 89. Schrijver, A. Theory of Linear and Integer Programming; John Wiley & Sons: Hoboken, NJ, USA, 1998; ISBN 0471982326.
- 90. Scales, L.E. Introduction to Non-Linear Optimization; Macmillan International Higher Education: London, UK, 1985; ISBN 1349177415.
- Xiao, Y.; Sun, W.; Sun, L. Dynamic Programming Based Economic Day-Ahead Scheduling of Integrated Tri-Generation Energy System with Hybrid Energy Storage. J. Energy Storage 2021, 44, 103395. [CrossRef]
- 92. Powell, W.B. *Approximate Dynamic Programming: Solving the Curses of Dimensionality;* John Wiley & Sons: Hoboken, NJ, USA, 2007; Volume 703, ISBN 0470182954.

- 93. Birge, J.R.; Louveaux, F. Introduction to Stochastic Programming; Springer Science & Business Media: New York, NY, USA, 2011; ISBN 1461402379.
- 94. Chen, X.; Sim, M.; Sun, P. A Robust Optimization Perspective on Stochastic Programming. *Oper. Res.* 2007, 55, 1058–1071. [CrossRef]
- 95. Nemirovski, A.; Juditsky, A.; Lan, G.; Shapiro, A. Robust Stochastic Approximation Approach to Stochastic Programming. *SIAM J. Optim.* **2009**, *19*, 1574–1609. [CrossRef]
- 96. Garcia, C.E.; Prett, D.M.; Morari, M. Model Predictive Control: Theory and Practice—A Survey. *Automatica* **1989**, *25*, 335–348. [CrossRef]
- 97. Allgöwer, F.; Zheng, A. Nonlinear Model Predictive Control; Birkhäuser: Basel, Switzerland, 2012; Volume 26, ISBN 3034884079.
- Kampelis, N.; Tsekeri, E.; Kolokotsa, D.; Kalaitzakis, K.; Isidori, D.; Cristalli, C. Development of Demand Response Energy Management Optimization at Building and District Levels Using Genetic Algorithm and Artificial Neural Network Modelling Power Predictions. *Energies* 2018, *11*, 3012. [CrossRef]
- Bot, K.; Ruano, A.; Ruano, M.d.G. Forecasting Electricity Consumption in Residential Buildings for Home Energy Management Systems. In Proceedings of the International Conference on Information Processing and Management of Uncertainty in Knowledge-Based Systems, Lisbon, Portugal, 15–19 June 2020; Springer: Berlin/Heidelberg, Germany, 2020; pp. 313–326.
- 100. Bot, K.; Santos, S.; Laouali, I.; Ruano, A.; Ruano, M.d.G. Design of Ensemble Forecasting Models for Home Energy Management Systems. *Energies* **2021**, *14*, 7664. [CrossRef]
- 101. Koltsaklis, N.; Panapakidis, I.P.; Pozo, D.; Christoforidis, G.C. A Prosumer Model Based on Smart Home Energy Management and Forecasting Techniques. *Energies* **2021**, *14*, 1724. [CrossRef]
- 102. Bot, K.; Ruano, A.; Ruano, M.d.G. Short-Term Forecasting Photovoltaic Solar Power for Home Energy Management Systems. *Inventions* **2021**, *6*, 12. [CrossRef]
- Fontenot, H.; Dong, B. Modeling and Control of Building-Integrated Microgrids for Optimal Energy Management—A Review. *Appl. Energy* 2019, 254, 113689. [CrossRef]
- 104. Tavakoli, M.; Shokridehaki, F.; Marzband, M.; Godina, R.; Pouresmaeil, E. A Two Stage Hierarchical Control Approach for the Optimal Energy Management in Commercial Building Microgrids Based on Local Wind Power and PEVs. *Sustain. Cities Soc.* 2018, 41, 332–340. [CrossRef]
- 105. Pearl, J. Heuristics: Intelligent Search Strategies for Computer Problem Solving; Addison-Wesley Longman Publishing Co., Inc.: Boston, MA, USA, 1984.
- Rothlauf, F. Design of Modern Heuristics: Principles and Application; Springer Science & Business Media: New York, NY, USA, 2011; ISBN 3540729623.
- 107. Osman, I.H.; Kelly, J.P. Meta-Heuristics Theory and Applications. J. Oper. Res. Soc. 1997, 48, 657. [CrossRef]
- Jones, D.F.; Mirrazavi, S.K.; Tamiz, M. Multi-Objective Meta-Heuristics: An Overview of the Current State-of-the-Art. *Eur. J. Oper. Res.* 2002, 137, 1–9. [CrossRef]
- Lee, S.; Choi, D.-H. Energy Management of Smart Home with Home Appliances, Energy Storage System and Electric Vehicle: A Hierarchical Deep Reinforcement Learning Approach. Sensors 2020, 20, 2157. [CrossRef]
- Dupačová, J.; Gröwe-Kuska, N.; Römisch, W. Scenario Reduction in Stochastic Programming. *Math. Program.* 2003, 95, 493–511.
   [CrossRef]
- 111. Heitsch, H.; Römisch, W. Scenario Reduction Algorithms in Stochastic Programming. *Comput. Optim. Appl.* **2003**, *24*, 187–206. [CrossRef]
- 112. Dashtdar, M.; Bajaj, M.; Hosseinimoghadam, S.M.S. Design of Optimal Energy Management System in a Residential Microgrid Based on Smart Control. *Smart Sci.* 2021, *10*, 25–39. [CrossRef]
- 113. Terlouw, T.; AlSkaif, T.; Bauer, C.; van Sark, W. Optimal Energy Management in All-Electric Residential Energy Systems with Heat and Electricity Storage. *Appl. Energy* **2019**, *254*, 113580. [CrossRef]
- 114. Govindan, K.; Fattahi, M.; Keyvanshokooh, E. Supply Chain Network Design under Uncertainty: A Comprehensive Review and Future Research Directions. *Eur. J. Oper. Res.* **2017**, *263*, 108–141. [CrossRef]
- Fiorini, L.; Aiello, M. Energy Management for User's Thermal and Power Needs: A Survey. *Energy Rep.* 2019, 5, 1048–1076.
   [CrossRef]
- Parrish, B.; Heptonstall, P.; Gross, R.; Sovacool, B.K. A Systematic Review of Motivations, Enablers and Barriers for Consumer Engagement with Residential Demand Response. *Energy Policy* 2020, 138, 111221. [CrossRef]
- 117. Khan, A.R.; Mahmood, A.; Safdar, A.; Khan, Z.A.; Khan, N.A. Load Forecasting, Dynamic Pricing and DSM in Smart Grid: A Review. *Renew. Sustain. Energy Rev.* 2016, 54, 1311–1322. [CrossRef]
- Ali, S.; Zheng, Z.; Aillerie, M.; Sawicki, J.-P.; Péra, M.-C.; Hissel, D. A Review of DC Microgrid Energy Management Systems Dedicated to Residential Applications. *Energies* 2021, 14, 4308. [CrossRef]
- 119. Bui, V.-H.; Hussain, A.; Im, Y.-H.; Kim, H.-M. An Internal Trading Strategy for Optimal Energy Management of Combined Cooling, Heat and Power in Building Microgrids. *Appl. Energy* **2019**, 239, 536–548. [CrossRef]
- 120. Fang, X.; Wang, J.; Song, G.; Han, Y.; Zhao, Q.; Cao, Z. Multi-Agent Reinforcement Learning Approach for Residential Microgrid Energy Scheduling. *Energies* **2020**, *13*, 123. [CrossRef]
- Farinis, G.K.; Kanellos, F.D. Integrated Energy Management System for Microgrids of Building Prosumers. *Electr. Power Syst. Res.* 2021, 198, 107357. [CrossRef]

- Arcos-Aviles, D.; Pascual, J.; Marroyo, L.; Sanchis, P.; Guinjoan, F. Fuzzy Logic-Based Energy Management System Design for Residential Grid-Connected Microgrids. *IEEE Trans. Smart Grid* 2016, *9*, 530–543. [CrossRef]
- Islam, M.; Yang, F.; Ekanayek, C.; Amin, M. Grid Power Fluctuation Reduction by Fuzzy Control Based Energy Management System in Residential Microgrids. *Int. Trans. Electr. Energy Syst.* 2019, 29, e2758. [CrossRef]
- 124. Tooryan, F.; HassanzadehFard, H.; Collins, E.R.; Jin, S.; Ramezani, B. Optimization and Energy Management of Distributed Energy Resources for a Hybrid Residential Microgrid. *J. Energy Storage* **2020**, *30*, 101556. [CrossRef]
- 125. Ali, S.; Malik, T.N.; Raza, A. Hybrid Risk-averse Energy Management Optimizer for Large-scale Industrial Building Microgrids. Int. Trans. Electr. Energy Syst. 2020, 30, e12442. [CrossRef]
- 126. Dagdougui, Y.; Ouammi, A.; Benchrifa, R. Energy Management-Based Predictive Controller for a Smart Building Powered by Renewable Energy. *Sustainability* **2020**, *12*, 4264. [CrossRef]
- 127. Eseye, A.T.; Lehtonen, M.; Tukia, T.; Uimonen, S.; Millar, R.J. Optimal Energy Trading for Renewable Energy Integrated Building Microgrids Containing Electric Vehicles and Energy Storage Batteries. *IEEE Access* 2019, 7, 106092–106101. [CrossRef]
- 128. Thomas, D.; Deblecker, O.; Ioakimidis, C.S. Optimal Operation of an Energy Management System for a Grid-Connected Smart Building Considering Photovoltaics' Uncertainty and Stochastic Electric Vehicles' Driving Schedule. *Appl. Energy* 2018, 210, 1188–1206. [CrossRef]
- Vincent, R.; Houari, A.; Ait-Ahmed, M.; Benkhoris, M.F. Influence of Different Time Horizon-Based Battery Energy Management Strategies on Residential Microgrid Profitability. J. Energy Storage 2020, 29, 101340. [CrossRef]
- 130. Ruano, A. Non-Intrusive Load Monitoring Applied to Intelligent Home Energy Management Systems. 2022. Available online: https://csi.ualg.pt/nilmforihem (accessed on 31 November 2021).